REGIONAL PARK/SPORTS COMPLEX 1-STEP DESIGN-BUILD CONTRACT

THIS AGREEMENT, made and entered by and between CITY OF MARICOPA, an Arizona municipal corporation, hereinafter designated the "CITY" and HAYDON BUILDING CORP, hereinafter designated the "DESIGN-BUILDER."

RECITALS

A. The City Council of the City of Maricopa, Arizona, is authorized and empowered by provisions of the City Code to execute contracts for professional services and construction services.

B. The City intends to construct the Regional Park/Sports Complex located South of Bowlin Road and East of John Wayne Parkway (Section 34, Township 4S, Range 3E) herein after referred to as the "Project". Such Project is specifically described in Exhibit A, which is attached hereto and incorporated into this Agreement by reference.

C. In response to a Request for Statement of Qualifications, the Design-Builder has represented to the City the ability to design and construct the Project and based on this representation the City has engaged Haydon Building Corporation to design and construct the Project. The Request for Statement of Qualifications and Design-Builder's response is attached hereto as Exhibit B, which is incorporated into this Agreement by reference.

NOW THEREFORE, for and in consideration of the mutual covenants and considerations hereinafter contained, it is agreed by and between the City and the Design-Builder as follows:

ARTICLE 1 - DEFINITIONS

"Addenda" written or graphic instruments issued prior to the submittal of the GMP Proposal(s), which clarify, correct or change the GMP Proposal(s) requirements.

"Agreement" or "Contract" means this written document signed by the City and the Design-Builder covering the design and construction of the Project, and including other documents itemized and referenced in or attached to and made part of this Agreement.

"Change Order" (Amendment) means a written instrument issued after execution of this Agreement signed by the City and the Design-Builder, stating their agreement upon all of the following: the scope of the change in the Work; the amount of the adjustment to the Contract Price; the extent of the adjustment to the Contract Time(s) or modifications of other contract terms.

"City" (Owner) means the City of Maricopa, a municipal corporation, with whom the Design-Builder has entered into this Agreement and for whom the services is to be provided pursuant to said Agreement.

"City's Representative" means the City's Program Manager, ABACUS Project Management, an Arizona Corporation, the designee is included in Subsection 8.4.1.2.

"City's Senior Representative" means the person designated in Subsection 8.4.1.1.

"Commissioning" means the process prescribed in Section 2.21 for achieving, validating and documenting the performance of the total building and its systems to meet the design needs and requirements of the City.

"Construction Documents" means the plans, specifications and drawings prepared by the Design-Build team and includes documents as prescribed by Section 2.8.

"Construction Drawings" means the detailed drawings approved as part of the Construction Documents.

"Construction Fee" means the Design-Builder's administrative costs, home office overhead, and profit.

"Contract Documents" means the following items and documents in descending order of precedence executed by the City and the Design-Builder: (i) all written modifications, amendments and Change Orders signed by both parties (ii) this Agreement, including all exhibits and attachments; (iii) Construction Documents; and (iv) Design-Builder's GMP Proposal.

"Contract Price" means the amount or amounts set forth in Article 5 subject to adjustment in accordance with this Agreement.

"Contract Time" means the period of time, as set forth in Article 4, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.

"Cost of the Work" means the direct costs necessarily incurred by the Design-Builder in the proper performance of the Work. The Cost of the Work shall include direct labor costs, subcontract costs, costs of materials and equipment incorporated in the completed construction, costs of other materials and equipment, temporary facilities, building permit fees (if not paid for by City), materials testing, and related items. The Cost of the Work shall not include the Design-Builder's Construction Fee, General Conditions Costs, or taxes.

"Critical Path" means the sequence of activities from the start of the Work to the Substantial Completion of the Project. Any delay in the completion of these activities will extend the Substantial Completion date.

"Day(s)" means calendar days unless otherwise specifically noted in the Contract Documents.

"Deliverables" means the work products prepared by the Design-Builder in performing the scope of work described in this Contract. Some of the major deliverables to be prepared and provided by the Design-Builder during the design phase may include but are not limited to: Construction Management Plan, Project Schedule, Schedule of Values, alternative system evaluations, procurement strategies and plans, cost estimates, construction market surveys, cash flow projections, GMP Proposals, Subcontractor procurement plan, Subcontractor agreements, Subcontractor bid packages, Supplier agreements, and others as indicated in this Contract or required by the Project Team.

"Design Services" means all professional services to be performed or procured by the Design-Builder to provide required Project design under this Agreement and any subsequent amendments.

"Design-Builder" means the firm, corporation, or other approved legal entity with whom the City has entered into this Contract to provide services as detailed in this Contract.

"Design-Builder's Contingency" means a fund to cover cost growth during the Project used at the discretion of the Design-Builder usually for costs that result from Project circumstances. The amount of the Design-Builder's Contingency will be negotiated as a separate line item in each GMP package. Use and management of the Design-Builder's Contingency is described in Subsection 5.3.3.6.

"Design-Builder's Senior Representative" means the person designated in Subsection 8.4.2.1.

"Design-Builder's Representative" means the person designated in Subsection 8.4.2.2.

"Design Professional" means a qualified, licensed professional who furnishes Design Dervices required under the Contract Documents.

"Differing Site Conditions" means concealed or latent physical conditions or subsurface conditions at the Site that, (i) materially differ or are not reasonably inferable from the conditions indicated in the Contract Documents or (ii) are of an unusual nature, differing materially from the conditions ordinarily encountered and generally recognized as inherent in the Work.

"Final Acceptance" means the completion of the Project as prescribed in Section 4.4.

"Float" means the number of Days by which an activity can be delayed without lengthening the Critical Path and extending the Substantial Completion date.

"General Conditions Costs" includes, but is not limited to the following types of costs for the Design-Builder during the construction phase: (i) payroll costs for project manager or construction manager for Work conducted at the Site; (ii) payroll costs for the superintendent and full-time general foremen; (iii) payroll costs for other management personnel resident and working at the Site; (iv) workers not included as direct labor costs engaged in support (e.g. loading/unloading, clean-up, etc.); (v) administrative office personnel; (vi) costs of offices and temporary facilities including office materials, office supplies, office equipment, and minor expenses; (vii) utilities, fuel, sanitary facilities and telephone services at the Site; (viii) costs of liability insurance premiums not included in labor burdens for direct labor costs; (ix) costs of bond premiums; (x) cost of consultants not in the direct employ of the Design-Builder or Subcontractors; and (xi) fees for licenses.

"Guaranteed Maximum Price" or "GMP" means the sum of the cost for design, permitting and maximum cost of the construction including the Design-Builder's Construction Fee, General Conditions Costs; sales tax, and the Design-Builders Contingency.

"GMP Plans and Specifications" means the plans and specifications upon which the Guaranteed Maximum Price proposal is based.

"Legal Requirements" means all applicable federal, state and local laws, codes, ordinances, rules, regulations, orders and decrees of any government or quasi-government entity having jurisdiction over the Project or Site, the practices involved in the Project, Site or Work.

"Notice to Proceed" or "NTP" means the directive issued by the City, authorizing the Design-Builder to start Work.

"Owner's Contingency" means a fund to cover cost growth during the Project used at the discretion of the City usually for costs that result from the City's directed changes or unforeseen Site conditions. The amount of the Owner's contingency will be set solely by the City and will be in addition to the Project costs included in the Design-Builder's GMP packages. Use and management of the Owner's contingency is described in Section 5.3.4.

"Payment Request" means the City form used by the Design-Builder to request payment for Work in accordance with Article 7.

"Performance Period" means the period of time allotted in the Contract Documents to complete the construction comprised within a GMP. The Performance Period shall be stated with each GMP and shown on the Project Schedule.

"Product Data" means illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Design-Builder to illustrate materials or equipment for some portion of the construction.

"Project" means the Work to be completed in the execution of this Agreement and as amended and as described in the Recitals above and in Exhibit A.

"Project Schedule" means a schedule as prescribed in Section 2.4.

"Record Documents" means the documents created pursuant to Section 2.19.

"Samples" means physical examples which illustrate materials, equipment or workmanship and establish standards by which the construction will be judged.

"Schedule of Values (SOV)", means Document specified in the construction phase, which divides the Contract Price into pay items, such that the sum of all pay items equals the Contract Price for the construction phase Work, or for any portion of the Work having a separate specified Contract Price.

"Shop Drawings" mean drawings, diagrams, schedules and other data specially prepared for the construction by the Design-Builder or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the construction.

"Site" means the land or premises on which the Project is located.

"Subcontractor" or "Subconsultant" means an individual or firm having a direct contract with the Design-Builder or any other individual or firm having a contract with the aforesaid contractors at any tier, who undertakes to perform a part of the construction phase Work for which the Design-Builder is responsible.

"Substantial Completion" means when the Work, or an agreed upon portion of the Work, is sufficiently complete so that the City can occupy and use the Project or a portion thereof for its intended purposes. This may include, but is not limited to: (i) approval by the Fire Marshall, Building Official and local authorities (Certificate of Occupancy); (ii) Elevator Permit; (iii) all systems in place, functional, and displayed to the City or its representative; (iv) all materials and equipment installed; (v) all systems reviewed and accepted by the City; (vi) draft O&M manuals and record documents reviewed and accepted by the City; (vii) City operation and maintenance training complete; (viii) HVAC test and balance completed (Provide minimum 30 days prior to projected substantial completion); (ix) landscaping and Site work; and (x) final cleaning. The conditions of Substantial Completion that do not apply to a specific GMP will be listed in the Notice to Proceed Letter.

"Supplier" means a manufacturer, fabricator, supplier, distributor, material man or vendor having a direct contract with the Design-Builder or any Subcontractor to furnish materials or equipment to be incorporated in the construction phase Work by the Design-Builder or any Subcontractor.

"Work" or "construction" means all the Design-Builder's design, any construction and other services required by the Contract Documents, including procuring and furnishing materials, equipment, services, and labor reasonably inferable from the Contract Documents.

ARTICLE 2 – DESIGN-BUILDER'S SERVICES AND RESPONSIBILITIES

2.0 The following Sections prescribe the services and responsibilities required for the proper execution and completion of the Work by the Design-Builder. They are not organized in any specific order and may pertain to all phases of the Work.

2.1 General Services

- **2.1.1** The Design-Builder shall, through personnel employed by the Design-Builder, or procured from qualified Subconsultants or Subcontractors, perform all Work, and provide all material, equipment, tools and labor, necessary to complete the Work described in and reasonably inferable from this Agreement.
- **2.1.1.1** This Agreement shall be executed and signed by the City and the Design-Builder, stating their agreement to proceed with design in accordance with all codes, standards and requirements as adopted by ordinance or as identified in the design/preconstruction scope & fee included in this Article 2 and Exhibit C.
- **2.1.1.2** An amendment to this Agreement may be executed and signed by the City and the Design-Builder, stating their agreement to a GMP and to proceed with the completion of the design and construction of the Project.
- **2.1.2** The City hereby acknowledges its acceptance of the key personnel of the design-build team as submitted by the Design-Builder in its statement of qualifications or as amended subsequently

during the selection process. At any time hereafter that the Design-Builder desires to reassign or change key personnel while performing under this Agreement, the Design-Builder shall submit a request to reassign or change key personnel and the qualifications of the proposed new key personnel to the City for prior approval.

- **2.1.3** The Design-Builder will maintain an adequate and competent staff of qualified persons, as may be determined by the City, throughout the performance of this Agreement. If the City objects, with reasonable cause, to any of the Design-Builder's design professionals, the Design-Builder shall take prompt corrective action acceptable to the City and, if required, remove such persons from the Project and replace with new design professionals agreed to by the City.
- 2.1.4 The Design-Builder's Representative shall be reasonably available to the City and shall have the necessary expertise and experience required to supervise the Work. The Design-Builder's Representative shall communicate regularly with the City and shall be vested with the authority to act on behalf of the Design-Builder. The Design-Builder's Representative may be replaced as described in Section 2.1.2.
- **2.1.5** The parties will meet within seven (7) days after execution of this Agreement to discuss issues affecting the administration of the Work, the Project Schedule and to implement the necessary procedures, including those relating to submittals and payment, to facilitate the ability of the parties to perform their obligations under this Agreement. The Notice to Proceed date prescribed in Section 4.1 will be established.
- **2.1.6** The Design-Builder shall coordinate their Work with the City's Designer, Architekton and CM@R, CORE Construction for the Multigenerational Center/Aquatic Facility which is located on the same site.

2.2 Professional Services

- **2.2.1** The Design-Builder shall, consistent with applicable state licensing laws, provide through qualified, licensed design professionals employed by the Design-Builder, or procured from qualified, independent licensed design consultants, the necessary Design Services, including architectural, engineering and other design professional services, for the preparation of the required drawings, specifications and other design submittals to permit the Design-Builder to complete the Work consistent with the Contract Documents. The Design-Builder's design professionals shall seal with an Arizona registered professional seal all plans, works, and deliverables prepared by them for this Agreement as required by state law.
- 2.2.2 The Design-Builder understands and agrees that the Senior City Representative or the City Representative shall be the sole contact for administering this Agreement. The Design-Builder is not precluded from discussing the Project, or its requirements with the tenants or other entities which will ultimately use the facility, but all specific directions to or requests of the Design-Builder must be authorized by the Senior City Representative or the City Representative.

2.3 Standard of Care for Design Professional Services and Corrections

- **2.3.1** The standard of care for all design professional services performed to execute the Work shall be the care and skill ordinarily used by members of the design profession practicing under similar conditions at the same time and locality of the Project. Notwithstanding the preceding sentence, if the parties agree upon specific performance standards for any aspect of the services, which standards are to be set forth in an exhibit to this Agreement entitled "Performance Standard Requirements," the design professional services shall be performed to achieve such standards.
- **2.3.2** The Design-Builder shall be responsible for the completeness and accuracy of the plans, specifications, supporting data, and other work prepared or compiled under its obligation for this Project and shall correct, at its expense, all willful or negligent errors, omissions and acts therein which may be discovered. Correction of willful or negligent errors, omissions and acts

discovered on architectural or engineering plans and specifications shall be the responsibility of the Design-Builder. The cost of the design necessary to correct those errors attributable to the Design-Builder shall not be reimbursable costs to the Design-Builder. Any damage incurred by the City as a result of additional construction cost caused by such willful or negligent errors, omissions or acts shall not be reimbursed to the Design-Builder to the extent that such willful or negligent errors, omissions and acts fall below the standard of care and skill that a registered professional in Arizona would exercise under similar conditions. The fact that the City has accepted or approved the Design-Builder's product shall in no way relieve the Design-Builder of any of its responsibilities.

2.4 Project Schedule

- **2.4.1** The Project Schedule shall be established, updated and maintained throughout the Work. An updated Project Schedule shall be part of the GMP amendment.
- 2.4.2 The Project Schedule shall be revised as required by conditions and progress of the Work, but such revisions shall not relieve the Design-Builder of its obligations to complete the Work within the Contract Time(s), as such dates may be adjusted in accordance with the Contract Documents.
- **2.4.3** An updated Project Schedule shall be submitted monthly to the City five (5) Days prior to the Design-Builder's monthly Payment Request as prescribed in Article 7.
- 2.4.3.1 The Design-Builder shall provide the City with a monthly status report with each Project Schedule detailing the progress of the Work, including whether (i) the Work is proceeding according to schedule, (ii) discrepancies, conflicts, or ambiguities exist in the Contract Documents that require resolution, and (iii) other items require resolution so as not to jeopardize the Design-Builder's ability to complete the Work for the Contract Price(s) and within the Contract Time(s).
- **2.4.3.2** With each Project Schedule submitted, the Design-Builder shall include a transmittal letter including the following.
 - Description of problem tasks (referenced to field instructions, requests for information (RFIs), change order or claim numbers) as appropriate.
 - Current and anticipated delays not resolved by approved change order, including:
 - Cause of the delay
 - o Corrective action and schedule adjustments to correct the delay
 - Known or potential impact of the delay on other activities, milestones, and by the date of Substantial Completion.
 - Changes in construction sequence
 - Pending items and status thereof including, but not limited to:
 - Pending change orders
 - Time extension requests
 - o Other items
 - Substantial Completion date status:
 - o If ahead of schedule, the number of Days ahead.
 - If behind schedule, the number of Days behind.
 - Other Project or scheduling concerns
- 2.4.4 The City's review of and response to the Project Schedule is only for general conformance with the scheduling requirements of the Contract Documents. The review shall not relieve the Design-Builder from compliance with the requirements of the Contract Documents or be construed as relieving the Design-Builder of its complete and exclusive control over the means, methods, sequences and techniques for executing the Work.

- **2.4.4.1** Upon the City's request, the Design-Builder shall participate in the review of the Design-Builder's Project Schedule submissions. The City may request the participation of Subconsultants and/or Subcontractors in these reviews, as determined necessary by the City.
- **2.4.5** The Project Schedule shall include a Critical Path Method (CPM) diagram schedule that shall show the sequence of activities, the interdependence of each activity and indicate the path of critical activities.
- **2.4.5.1** The CPM diagram schedule shall be in Days and indicate duration, earliest and latest start and finish dates, and total Float times for all activities except critical activities. The CPM diagram shall be presented in a time scaled graphical format for the Project as a whole.
- 2.4.5.2 The CPM diagram schedule shall indicate all relationships between activities.
- **2.4.5.3** The activities making up the schedule shall be of sufficient detail to assure that adequate planning has been done for proper execution of the Work and such that it provides an appropriate basis for monitoring and evaluating the progress of the Work.
- **2.4.5.4** The CPM diagram construction schedule shall be based upon activities, which would coincide with the Schedule of Values.
- **2.4.5.5** The CPM diagram schedule shall show all submittals associated with each work activity and the review time for each submittal.
- **2.4.5.6** The schedule shall show milestones, including milestones for Owner-furnished information, and shall include activities for Owner-furnished equipment and furniture when those activities are interrelated with the Design-Builder's activities.
- **2.4.5.7** The Project Schedule shall consider the City's and the tenants' occupancy requirements showing portions of the Project having occupancy priority, and Contract Time.
- 2.4.6 Float time shall be as prescribed below:
- **2.4.6.1** The total Float within the overall schedule, is not for the exclusive use of either the City or the Design-Builder, but is jointly owned by both and is a resource available to and shared by both parties as needed to meet contract milestones and the Project completion date.
- **2.4.6.2** The Design-Builder shall not sequester shared Float through such strategies as extending activity duration estimates to consume available Float, using preferential logic, or using extensive crew/resource sequencing, etc. Since Float time within the schedule is jointly owned, no time extensions will be granted nor delay damages paid until a delay occurs which extends the Work beyond the Substantial Completion date.
- **2.4.6.3** Since Float time within the construction schedule is jointly owned, it is acknowledged that Citycaused delays on the Project may be offset by City-caused time savings (i.e., critical path submittals returned in less time than allowed by the Contract, approval of substitution requests and credit changes which result in a savings of time to the Design-Builder, etc.). In such an event, the Design-Builder shall not be entitled to receive a time extension or delay damages until all City-caused time savings are exceeded, and the Substantial Completion date is also exceeded.

2.5 Cost Estimates

- **2.5.1** The Project budget available for the Work will be communicated to the Design-Builder through separate correspondence. The Design-Builder is responsible for the delivery of the Project covered by the Contract Documents within the Project budget.
- **2.5.2** With each Project Schedule submittal pursuant Section 2.4.3, the Design-Builder shall provide a detailed cost estimate and a written review of the documents.

- **2.5.2.1** If any estimate submitted to the City exceeds previously accepted estimates or the City's Project budget, the Design-Builder shall make at its sole expense appropriate recommendations on methods and materials to the City that he believes will bring the project back into the Project budget.
- 2.5.2.2 In between these milestone estimates, the Design-Builder shall periodically provide a tracking report which identifies the upward or downward movements of costs due to value engineering or scope changes. It shall be the responsibility of the Design-Builder to keep the City informed as to the major trend changes in costs relative to the City's budget.
- **2.5.3** If requested by the City, the Design-Builder shall prepare a preliminary "cash flow" projection based upon historical records of similar type projects to assist the City in the financing process.

2.6 Intentionally Omitted

2.7 Construction Management Plan

- **2.7.1** The Design-Builder will prepare a Construction Management Plan (CMP).
- 2.7.2 The CMP shall include:
 - * Project milestone dates and the Project Schedule, including the broad sequencing of the design and construction of the Project,
 - * Investigations, if any, to be undertaken to ascertain subsurface conditions and physical conditions of existing surface and subsurface facilities and underground utilities,
 - * Alternate strategies for fast-tracking and/or phasing the construction,
 - * Goal compliance strategy,
 - * The number of separate sub-agreements to be awarded to Subcontractors and Suppliers for the Project construction,
 - * Permitting strategy,
 - * Safety and training programs,
 - * Construction quality control,
 - * Commissioning program,
 - * Cost estimate and basis of the model, and
 - * Matrix summarizing each Project Team member's responsibilities and roles.
- **2.7.3** The Design-Builder shall add detail to its previous version of the CMP to keep it current throughout the design phase, so that the CMP is ready for implementation at the start of construction.
- **2.7.4** The Design-Builder shall submit the initial CMP and subsequent updated versions of the CMP to the City in electronic format CD-ROM.

2.8 Design Services

- **2.8.1** The Design-Builder shall provide all interim design submissions and deliverables as proscribed in the design standards and guidelines provided by the City, and as shown on the Project Schedule.
- **2.8.1.1** Within seven (7) Days after a scheduled submission, the Design-Builder and the City shall meet and confer about the submissions, with the Design-Builder identifying during such meetings, among other things, the evolution of the design and any significant changes or deviations from the Contract Documents, or previously submitted design submissions.

- **2.8.1.2** The Design-Builder, with each required submittal, shall submit and distribute four (4) hard copy sets of plans and specifications and one (1) set of plans in AutoCAD format compatible with City of Maricopa CADD technology.
- **2.8.1.3** Minutes of the meetings will be maintained by the Design-Builder and provided within five (5) Days following the design review meeting to all attendees for review.
- **2.8.1.4** The City shall review and approve the interim design submissions in a time that is consistent with the turnaround times set forth in the Design-Builder's Project Schedule.
- **2.8.1.5** The Design-Builder shall not cause the design to proceed until the City approves the interim design submissions as prescribed in this Section. If the Design-Builder allows the design to proceed without City approval, the cost of any resultant redesign is not a reimbursable cost.
- **2.8.1.6** The City's review and approval of interim design submissions and the Construction Documents is for the purpose of mutually establishing a conformed set of Construction Documents compatible with the requirements of the Project. Neither the City's review nor approval of any interim design submissions and Construction Documents shall be deemed to transfer any design liability to the City.
- 2.8.2 The Project design must meet all applicable (i) Maricopa County Government (MAG) Uniform Standard Technical Specifications and Uniform Standard Details and Drawings, latest revision; (ii) all City of Maricopa Engineering Standards, MAG Supplements and the Pinal County Drainage Manual; (iii) all City building standards and City Codes; and (iv) shall include all special provisions provided by the City.
- **2.8.3** The Project design criteria and specifications shall be in accordance with all codes, standards and requirements as adopted by Ordinance. Variances from the standards, guidelines and review process documents must be identified in writing by the Design-Builder and approved by the City. The Design-Builder shall identify conflicts between the design standards and guidelines and the requirements in Section 2.8.2 above or Legal Requirements and shall obtain concurrence with resolution of the conflict. The design standards, guidelines and review process documents or approval of variances or resolution of conflicts shall not be deemed to transfer any design liability to the City.
- **2.8.4** The Design-Builder shall not specify any construction materials known to be hazardous or potentially hazardous, including asbestos, lead or any derivative thereof unless specifically approved in writing by the City.
- **2.8.5** The Design-Builder shall coordinate with private, public and City utilities (i.e., power, gas, water and wastewater, flood control, public and private communications, etc.) regarding standard utility issues and incorporate pertinent information in the plans. The Design-Builder shall provide a Site Plan to the City project manager incorporating pertinent utility information for confirmation of locations and conflict review.
- **2.8.5.1** When required, the Design-Builder shall identify proposed private and public utility easements. The Design-Builder shall prepare documentation and forward to the City project manager for coordination of utility review and consideration of the proposed easement. The Design-Builder shall secure the services of a Registered Land Surveyor to provide a sealed survey and legal description for any required Private and Public Utility Easement for this Project. The submittal shall comply with the City standard format.
- **2.8.6** The Design-Builder shall be responsible for scheduling, submitting, obtaining approval and retrieving of all required Construction Documents to the various required reviewing agencies.
- **2.8.7** The Design-Builder shall be responsible for scheduling, preparing necessary exhibits and making presentation to the Planning & Zoning Commission, City Council and any appropriate community meetings relative to the Project.

- **2.8.8** The Design-Builder shall submit to the City Construction Documents setting forth in detail drawings and specifications describing the requirements for construction.
- **2.8.8.1** The Construction Documents shall be consistent with the latest set of interim design submissions, as such submissions may have been modified in a design review meeting.
- **2.8.8.2** The Design-Builder shall provide the drawings in AutoCAD format compatible with City of Maricopa CADD technology.
- **2.8.8.3** The drawing format will be a 24" x 36" sheet size unless otherwise authorized in writing by the City.
- **2.8.8.4** The parties shall have a design review meeting to discuss, and the City shall review and approve, the Construction Documents in accordance with the procedures set forth in this Article.
- **2.8.8.5** Prior to commencement of construction, the Design-Builder shall submit to the City the following:
 - One (1) set of approved Construction Drawings in AutoCAD format on electronic media (CD-ROM);
 - Three (3) print sets of approved Construction Drawings and three (3) half-size sets; and
 - Three (3) sets of specifications.
- **2.8.9** To the extent not prohibited by Legal Requirements, the Design-Builder may arrange for interim design submissions and Construction Documents for a portion of the Project to permit construction to proceed on that portion prior to completion of the Construction Documents for the entire Project.

2.9 Intentionally Omitted

2.10 Government Approvals and Permits

- **2.10.1** The Design-Builder shall obtain all necessary permits, approvals, and licenses required for the prosecution of the Work by any government or quasi-government entity having jurisdiction over the Project. The Design-Builder is specifically reminded of the need to obtain the necessary environmental permits or file the necessary environmental notices (including NPDES, 404, Dust Control, and other permits).
- **2.10.2** Copies of these permits and notices must be provided to the City's Representative prior to starting the permitted activity. In the case of Fire Department permits, a copy of the application for permit shall also be provided to the City's Representative. This provision does not constitute an assumption by the City of an obligation of any kind for violation of said permit or notice requirements.
- **2.10.3** The City shall be responsible for City of Maricopa review and permit(s) fees for building and demolition permits. The City will also pay review fees for grading and drainage, water, sewer, and landscaping. The City shall also pay for utility design fees for permanent services.
- **2.10.4** The Design-Builder shall be responsible for all other permits and review fees not specifically listed in Section 2.10.3 above.
- **2.10.5** The Design-Builder is responsible for the cost of water meter(s), water and sewer taps and fire lines and taps and all utility fees including incremental billings until Substantial Completion. Arrangements for construction water is the Design-Builder's responsibility.

2.11 Subcontractor Selection

- **2.11.1** Subcontractors shall be selected by a combination of qualifications and price derived through competitive bidding.
- **2.11.1.1** Design-Builder shall employ only Subcontractors who are duly licensed in Arizona and qualified to perform the Work per the requirements of the Contract Documents.
- 2.11.2 The Design-Builder will develop Subcontractor interest, submit the names of a minimum of three (3) qualified Subcontractors selected pursuant to a qualifications based procedure, for each trade in the Project for approval by the City and solicit bids for the various construction categories. If there are not three (3) qualified Subcontractors available for a specific trade or there are extenuating circumstances warranting such, the Design-Builder may request approval by the City to submit less than three (3) names. Without prior approval by the City, no change in the City-approved Subcontractors will be allowed.
- 2.11.2.1 If the Design-Builder desires to self-perform certain portions of the construction, it will request to be one of the approved Subcontractor bidders for those specific bid packages. The Design-Builder's bid will be evaluated in accordance with the process identified below. If after selection of a Subcontractor, events warrant and the City concurs, the Design-Builder may self perform construction without bidding or re-bidding.
- **2.11.2.2** If the City objects to any nominated Subcontractor or to any nominated self-performed construction for good reason, the Design-Builder will nominate a substitute Subcontractor.
- **2.11.2.3** The Design-Builder will distribute drawings and specifications, and when appropriate, conduct a Pre-bid conference with prospective Subcontractors.
- **2.11.2.4** The Design-Builder shall receive, open, record and evaluate the bids. The apparent low bidders will be interviewed to determine the responsiveness of their proposals. In evaluating the responsiveness of bid proposals the Design-Builder, in addition to bid price, shall consider the following factors: past performance on similar projects, qualifications and experience of personnel assigned, quality management plan, approach or understanding of the Work to be performed and performance schedule to complete the work. The final evaluation of subcontractor bids will be done with the City Representative in attendance to observe and witness the process. The Design-Builder will resolve any Subcontractor bid withdrawal, protest or disqualification in connection with the award at no increase in the cost of the construction.
- **2.11.3** Upon completion of the Subcontractor selection process, the Design-Builder shall submit a summary report to the City of the entire Subcontractor selection process. The report will indicate, by bid process, all Subcontractors contacted to determine interest, the Subcontractors solicited, the bids received and costs negotiated, and the selected Subcontractors for each category of Work.
- **2.11.4** The selected Subcontractors will provide a Schedule of Values, which will be used to create the overall Project Schedule of Values.

2.12 General Construction Services

- **2.12.1** Unless otherwise provided in the Contract Documents to be the responsibility of the City or a separate contractor, the Design-Builder shall provide through itself or Subcontractors the necessary supervision, labor, taxes, inspection, testing, start-up, material, equipment, machinery, arrangements for temporary utilities and other temporary facilities to permit the Design-Builder to complete construction of the Project consistent with the Contract Documents.
- **2.12.2** The Design-Builder shall completely and totally construct the Project and install the material therein for the City, in a good and workmanlike and substantial manner. The work shall be to the satisfaction of the City and strictly pursuant to and in conformity with the Contract Documents as amended.
- **2.12.3** The Design-Builder shall perform all construction activities efficiently and with the requisite expertise, skill and competence to satisfy the requirements of the Contract Documents. The

Design-Builder shall at all times exercise complete and exclusive control over the means, methods, sequences and techniques of construction.

2.13 Pre-construction Conference

- **2.13.1** After approval of Construction Documents and prior to the commencement of any construction, the City's Representative will schedule a Pre-construction conference.
- **2.13.2** The purpose of this conference is to establish a working relationship between the Design-Builder, utility firms, and various City agencies. The agenda will include critical elements of the construction schedule, submittal schedule, cost breakdown of major lump sum items, the level of Record Project Documents required, Payment Requests and processing, coordination with the involved utility firms, and emergency telephone numbers for all representatives involved in the course of construction.
- **2.13.3** The construction start date will be concurred to. After the meeting a Notice to Proceed letter will be issued confirming the construction start date, duration of the construction and if applicable the Substantial Completion date. If a Substantial Completion date is established the conditions of the Substantial Completion will be listed.
- **2.13.4** The Design-Builder shall provide a Schedule of Values based on the categories used in the buy out of the construction, but not greater than the approved GMP, and identifying the construction contingency. The Schedule of Values will subdivide the construction into all items comprising the construction.
- **2.13.5** Minimum attendance by the Design-Builder shall be the Design-Builder's Representative, who is authorized to execute and sign documents on behalf of the firm, the job superintendent, and the Design-Builder's safety officer.

2.14 Supervision of Construction

- **2.14.1** The Design-Builder or the Design-Builder's Representative shall at all times be present at the Site when construction activities are taking place.
- **2.14.1.1** All elements of the construction, such as concrete work, pipe work, etc., shall be under the direct supervision of a foreman or his designated representative on the Site, who shall have the authority to take actions required to properly carry out that particular element of the construction.
- **2.14.1.2** In the event of noncompliance of this Section 2.14, the City may require the Design-Builder to stop or suspend the construction in whole or in part. Such suspension, due to the Design-Builder's noncompliance shall not be considered a basis for an increase in the Contract Price or extension of Contract Time.
- **2.14.2** Where the Contract Documents require that a particular product be installed and/or applied by an applicator approved by the manufacturer, it is the Design-Builder's responsibility to ensure the Subcontractor employed for such work is approved.
- **2.14.3** During construction the City may reasonably object to any Subcontractor and the Subcontractor shall be removed from the construction, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that the City's decision impacts the Design-Builder's cost and/or time of performance.
- 2.14.4 Any person employed by the Design-Builder or any Subcontractor who, in the opinion of the City, does not perform his work in a proper, skillful and safe manner or is intemperate or disorderly shall, at the written request of the City, be removed from the construction by the Design-Builder or Subcontractor employing such person, and shall not be employed again in any portion of construction without the written approval of the City. The Design-Builder or Subcontractor shall

hold the City harmless from damages or claims, which may occur in the enforcement of this Section.

- **2.14.5** The Design-Builder assumes responsibility to the City for the proper performance of the Work of Subcontractors and any acts and omissions in connection with such performance. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between the City and any Subcontractor, including but not limited to any third-party beneficiary rights.
- **2.14.6** The Design-Builder shall coordinate the activities of all Subcontractors. If the City performs other work on the Project or at the Site with separate contractors under the City's control, the Design-Builder agrees to reasonably cooperate and coordinate its activities with those of such separate contractors so that the Project can be completed in an orderly and coordinated manner without unreasonable disruption.

2.15 Control of Construction Site

- 2.15.1 Throughout all phases of construction, including suspension of Work, the Design-Builder shall keep the Site reasonably free from debris, trash and construction wastes to permit the Design-Builder to perform its construction services efficiently, safely and without interfering with the use of adjacent land areas. Upon Substantial Completion of the construction, or a portion of the construction, the Design-Builder shall remove all debris, trash, construction wastes, materials, equipment, machinery and tools arising from the construction or applicable portions thereof to permit the City to occupy the Project or a portion of the Project for its intended use.
- **2.15.2** The Design-Builder shall take whatever steps, procedures or means to prevent any dust nuisance due to construction operations. The dust control measures shall be maintained at all times to the satisfaction of the City and in accordance with the requirements of the Pinal County Environmental Health Department Rules and Regulations.
- **2.15.2.1** The Design-Builder shall identify a trained responsible staff member who shall implement, monitor and maintain dust control procedures and assure compliance with the Pinal County Environmental Health Department regulations. For construction sites larger than five (5) acres, a Dust Control Manager shall be on-site for a minimum of four (4) hours per day until the site attains stabilization.
- **2.15.2.2** The Design-Builder shall notify the City Engineers authorized representative within twenty four (24) hours when receiving notice of noncompliant conditions, issuance of citations or Notice of Violations by the regulatory agency. The Design-Builder shall indemnify and reimburse the City for any fine, penalty, or monetary sanction imposed on the City which is caused by failure to adhere to the Pinal County Environmental Health Department Rules and Regulations.
- **2.15.3** The Design-Builder shall maintain ADA and ANSI accessibility requirements during construction activities in an occupied building or facility. ADA and ANSI accessibility requirements shall include, but not be limited to, parking, building access, entrances, exits, restrooms, areas of refuge, and emergency exit paths of travel. The Design-Builder shall be responsible for the coordination of all work to minimize disruption to building occupants and facilities.
- **2.15.4** In the event of abnormal weather conditions, such as windstorms, rainstorms, etc., the Design-Builder shall immediately inspect the work Site and take all necessary actions to insure public access and safety are maintained.
- **2.15.5** Only materials and equipment, which are to be used directly in the construction, shall be brought to and stored on the Site by the Design-Builder. When equipment is no longer required for the construction, it shall be removed promptly from the Site. Protection of construction materials and equipment stored at the Site from weather, theft, damage and all other adversity is solely the responsibility of the Design-Builder.

2.16 Shop Drawings, Product Data and Samples

- **2.16.1** Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. The purpose of their submittal is to demonstrate for those portions of the construction for which submittals are required the way the Design-Builder proposes to conform to the information given and the design concept expressed in the Construction Documents.
- **2.16.2** The Design-Builder shall review, approve, verify, and submit to the City three (3) copies of each Shop Drawing, Product Data, Sample, and similar submittal required by the Construction Documents in accordance with the approved construction schedule as prescribed in Section 4.2 as to cause no delay in the Work or in the activities of the City or of separate contractors.
- **2.16.3** By approving, verifying and submitting Shop Drawings, Product Data, Samples and similar submittals, the Design-Builder represents that the Design-Builder has determined and verified materials, field measurements and field construction criteria related thereto, or will do so, and has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Construction Documents.
- 2.16.4 The Design-Builder shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the City's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Design-Builder has specifically informed the City in writing of such deviation at the time of submittal and the City has given written approval to the specific deviation.
- **2.16.5** When professional certification of performance criteria of materials, systems or equipment is required by the Contract Documents, the City shall be entitled to rely upon the accuracy and completeness of such calculations and certifications.

2.17 Quality Control, Testing and Inspection

- **2.17.1** All materials used in the construction shall be new and unused, unless otherwise noted, and shall meet all quality requirements of the Contract Documents.
- **2.17.2** All construction materials to be used on the construction or incorporated into the construction, equipment, plant, tools, appliances or methods to be used in the construction may be subject to the inspection and approval or rejection of the City. Any material rejected by the City shall be removed immediately and replaced in an acceptable manner.
- 2.17.3 The procedures and methods used to sample and test material will be determined by the City. Unless otherwise specified, samples and test shall be made in accordance with the following: Maricopa Association of Governments (MAG) Uniform Standard Technical Specifications and Uniform Standard Details and Drawings latest version, the City of Maricopa Supplements to MAG and the standard methods of AASHTO or ASTM.
- **2.17.4** The City will select a pre-qualified City or Independent Testing Laboratory and will pay for initial City Acceptance Testing.
- **2.17.4.1** When the first and subsequent tests indicate noncompliance with the Contract Documents, the cost associated with that noncompliance will be paid for by the Design-Builder. Construction contingency cannot be utilized for the cost of re-testing.
- **2.17.4.2** When the first and subsequent tests indicate noncompliance with the Contract Documents, all retesting shall be performed by the same testing agency.
- 2.17.5 The Design-Builder will cooperate with the selected testing laboratory and all others responsible

for testing and inspecting the Work and shall provide them access to the Site at all times.

- **2.17.6** At the option of the City, materials may be approved at the source of supply before delivery is started.
- **2.17.7** Code compliance testing and inspections required by codes or ordinances, or by a plan approval authority, and which are made by a legally constituted authority, shall be the responsibility of and shall be paid by the Design-Builder, unless otherwise provided in the Contract Documents.
- **2.17.8** The Design-Builder's convenience and quality control testing and inspections shall be the sole responsibility of the Design-Builder and paid by the Design-Builder.
- **2.18 Trade names and Substitutions.** Construction Document references to equipment, materials or patented processes by manufacturer, trade name, make or catalog number, unless indicated that no substitutions are permitted, substitute or alternate items may be permitted, subject to the following:
- **2.18.1** The substitution shall be submitted by the Design-Builder in writing to the City.
- **2.18.2** The Design-Builder shall certify that the substitution will perform the functions and achieve the results called for by the general design, be similar and of equal substance, and be suited to the same use as that specified.
- **2.18.3** The submittal shall state any required changes in the Construction Documents to adapt the design to the proposed substitution.
- **2.18.4** The submittal shall contain an itemized estimate of all costs and credits that will result directly and indirectly from the acceptance of such substitution, including cost of design, license fees, royalties, and testing. Also, the submittal shall include any adjustment in the Contract Time created by the substitution. Substitutions will only be considered if they do not extend Contract Time.
- **2.18.5** The Design-Builder if requested by the City shall submit Samples or any additional information that may be necessary to evaluate the acceptability of the substitution.
- **2.18.6** The City will make the final decision and will notify the Design-Builder in writing as to whether the substitution has been accepted or rejected. If accepted, the Design-Builder will cause the Construction Documents to be revised as approved at the Design-Builder's cost.
- **2.18.7** The Design-Builder shall continue to perform the construction in accordance with the Construction Documents unless the City accepts the substitution and there is no extension of Contract Time.

2.19 Project Record Documents

- **2.19.1** During the construction period, the Design-Builder shall maintain at the jobsite a set of blueline or blackline prints of the Construction Drawings and shop drawings for Project Record Document purposes.
- **2.19.1.1** The Design-Builder shall mark these drawings to indicate the actual installation where the installation varies appreciably from the original Construction Drawings. Give particular attention to information on concealed elements, which would be difficult to identify or measure and record later. Items required to be marked include, but are not limited to:
 - Dimensional changes to the drawings.
 - Revisions to details shown on drawings.
 - Depths of foundations below first floor.
 - Locations and depths of underground utilities.

- Revisions to routing of piping and conduits.
- Revisions to electrical circuitry.
- Actual equipment locations.
- Duct size and routing.
- Locations of concealed internal utilities.
- Changes made by Change Order, Field Order, Report of Field Changes, ASI or RFI.
- Details not on original Construction Drawings.
- **2.19.1.2** The Design-Builder shall mark completely and accurately Project Record Drawing prints of Construction Drawings or shop drawings, whichever is the most capable of indicating the actual physical condition. Where shop drawings are marked, show cross-reference on Construction Drawings location.
- **2.19.1.3** The Design-Builder shall mark Project Record Drawings sets with red erasable colored pencil.
- **2.19.1.4** The Design-Builder shall note RFI Numbers, ASI Numbers and Change Order numbers, etc, as required to identify the source of the change to Construction Drawings.
- **2.19.1.5** The Design-Builder shall, as a condition of Substantial Completion, submit Project Record Drawing prints and shop drawings to the City or its representative for review and comment.
- **2.19.2** Final Project Record Drawings: Upon receipt of the reviewed Project Record Drawings from the City, the Design-Builder shall correct any deficiencies and/or omissions to the drawings and prepare the following for submission to the City within fourteen (14) Days:
- **2.19.2.1** On electronic media: (i) a complete set of Project Record Drawings prepared in Autocad format compatible with City of Maricopa's AutoCAD software, each drawing shall be clearly marked with "As-Built Document;" (ii) a complete set of project specifications in Microsoft Word format; and (iii) Shop Drawings in CADD or graphic file format.
- **2.19.2.2** On hard copy media: (i) a complete set of reproducible mylars from the final AutoCAD drawings in the format compatible with City of Maricopa's AutoCAD software; and (ii) five (5) edge bound blueline or blackline sets reproduced from these mylars.
- 2.19.2.3 The original copy of the Project Record Drawings (redline mark-ups).

2.20 Project Safety

- 2.20.1 The Design-Builder recognizes the importance of performing the Work in a safe manner so as to prevent damage, injury or loss to (i) all individuals at the Site, whether working or visiting, (ii) the Work, including materials and equipment incorporated into the Work or stored on-Site or off-Site, and (iii) all other property at the Site or adjacent thereto.
- **2.20.2** The Design-Builder assumes responsibility for implementing and monitoring all safety precautions and programs related to the performance of the Work.
- **2.20.3** The Design-Builder shall, prior to commencing construction, designate a Safety Representative with the necessary qualifications and experience to supervise the implementation and monitoring of all safety precautions and programs related to the Work. Unless otherwise required by the Contract Documents, the Design-Builder's Safety Representative shall be an individual stationed at the Site who may have other responsibilities on the Project in addition to safety.
- **2.20.4** The Safety Representative shall make routine daily inspections of the Site and shall hold weekly safety meetings with the Design-Builders personnel, subcontractors and others as applicable.

- **2.20.5** The Safety Representative shall hold weekly safety meetings with the Design-Builder's personnel, all Subcontractors of all tiers and others as applicable. Meetings shall be documented and minutes shall be recorded and made available to the City upon request.
- **2.20.6** The Design-Builder and Subcontractors of all tiers shall comply with all Legal Requirements relating to safety, as well as any City-specific safety requirements set forth in the Contract Documents, provided that such City-specific requirements do not violate any applicable Legal Requirement.
- 2.20.7 The Design-Builder will immediately report in writing any safety-related injury, loss, damage or accident arising from the Work to the City's Representative and, to the extent mandated by Legal Requirements, to all government or quasi-government authorities having jurisdiction over safety-related matters involving the Project or the Work. The Design-Builder shall establish a process for reporting, recording and abating all safety concerns or violations in a timely manner.
- **2.20.8** The Design-Builder's responsibility for safety under this Section is not intended in any way to relieve Subcontractors and Sub-Subcontractors of their own contractual and legal obligations and responsibility for (i) complying with all Legal Requirements, including those related to health and safety matters, and (ii) taking all necessary measures to implement and monitor all safety precautions and programs to guard against injury, losses, damages or accidents resulting from their performance of the Work.
- **2.20.8.1** The City acknowledges that Subcontractors are the controlling employer for the purposes of OSHA standards, enforcement and violations, other than Work self-performed by the Design-Builder.
- **2.20.8.2** The Design-Builder acknowledges that the City is not the controlling employer for the purpose of OSHA standards, enforcement and violations. The City of Maricopa, its employees, agents and representatives assume no responsibility of liability for the physical condition of the Project Site or the safety program for the Project.
- **2.20.9** The City reserves the right to conduct a contract safety compliance audit at any time during the term of the Project to verify the Design-Builder is in compliance with the terms and conditions of this Contract section as well as all applicable Legal Requirements relating to safety.

2.21 Commissioning

- **2.21.1** The Design-Builder shall develop a Commissioning program as part of its CMP pursuant to Section 2.7.
- **2.21.2** The Commissioning program shall include the roles and responsibilities of the City, tenants, and the Design-Builder.
- **2.21.3** The Design-Builder will provide a final Commissioning report indicating that the Work is functioning as prescribed in the Contract Documents and training completed as required by the Contract Documents.

2.22 Design-Builder's Warranty

- **2.22.1** The Design-Builder warrants to the City that the construction, including all materials and equipment furnished as part of the construction, shall be new unless otherwise specified in the Contract Documents, of good quality, in conformance with the Contract Documents and free of defects in materials and workmanship.
- **2.22.2** The Design-Builder's warranty obligation excludes defects caused by abuse, alterations, or failure to maintain the Project by persons other than the Design-Builder or anyone for whose acts the Design-Builder may be liable.
- 2.22.3 The Design-Builder's warranty obligation shall be for one (1) year, except for such greater

periods as may be required by the technical specifications.

- **2.22.4** Nothing in this warranty is intended to limit any manufacturer's warranty which provides the City with greater warranty rights than set forth in this Section or the Construction Documents.
- **2.22.5** The Design-Builder will provide the City with all manufacturers' warranties upon Substantial Completion.

2.23 Correction of Defective Construction

- **2.23.1** The Design-Builder agrees to correct any construction that is found to not be in conformance with the Construction Documents, including that part of the construction subject to Section 2.22 hereof, within a period of one (1) year from the date of Substantial Completion of the construction or any portion of the construction, or within such longer period to the extent required by the Contract Documents or provided by the Design-Builder or his Subcontractors. A progress payment, or partial or entire use or occupancy of the Project by the City shall not constitute acceptance of construction not in accordance with the Construction Documents.
- **2.23.2** The Design-Builder shall, within seven (7) Days of receipt of written notice from the City that the construction is not in conformance with the Construction Documents, take meaningful steps to commence correction of such nonconforming construction, including the correction, removal or replacement of the nonconforming construction and any damage caused to other parts of the construction affected by the nonconforming construction.
- **2.23.3** If the Design-Builder fails to commence the necessary steps within such seven (7) Day period, the City, in addition to any other remedies provided under the Contract Documents, may provide the Design-Builder with written notice that the City will commence correction of such nonconforming construction with its own forces. If the City does perform such corrective construction, the Design-Builder shall be responsible for all reasonable costs incurred by the City in performing such correction. Any such work performed by the City as a result of the Design-Builder's failure to commence corrective action shall not impact, in any way, the remaining warranty of the affected construction.
- **2.23.4** If the nonconforming construction creates an emergency requiring an immediate response, the Design-Builder will respond and initiate corrections within twenty-four (24) hours.
- **2.23.5** The one (1) year period referenced in Section 2.23.1 above applies only to the Design-Builder's obligation to correct nonconforming construction and is not intended to constitute a period of limitations for any other rights or remedies the City may have regarding the Design-Builder's other obligations under the Contract Documents.

ARTICLE 3 - CITY'S SERVICES AND RESPONSIBILITIES

3.1 Duty to Cooperate

- **3.1.1** The City shall, throughout the performance of Work, cooperate with the Design-Builder and perform its responsibilities, obligations and services in a timely manner to facilitate the Design-Builder's timely and efficient performance of the Work and so as not to delay or interfere with the Design-Builder's performance of its obligations under the Contract Documents.
- **3.1.2** The City shall provide timely reviews and approvals of interim design submissions and Construction Documents consistent with the turnaround times set forth in the Project Schedule.
- **3.1.3** The City's Representative as identified in Subsection 8.4.1.2 shall be responsible for providing City supplied information and approvals in a timely manner to permit the Design-Builder to fulfill its obligations under the Contract Documents. The City's Representative shall also provide the Design-Builder with prompt notice if it observes any failure on the part of the Design-Builder to

fulfill its contractual obligations, including any errors, omissions or defects in the performance of the Work.

- **3.1.4** Appropriate City Staff and tenants will be available and will participate in required training as part of the Commissioning activities.
- **3.1.5** If requested by the Design-Builder, the City's Representative will provide assistance and guidance in obtaining necessary permits. Regulating agencies of the City, such as Planning & Development Services and Fire Departments, enforce Legal Requirements. These enforcement activities are not subject to the responsibilities of the City under this Agreement.

3.2 Furnishing of Services and Information

- **3.2.1** The City will be responsible for the payment of the following:
- **3.2.1.1** City review and permit(s) fees for building and demolition permits.
- 3.2.1.2 City review fees for grading and drainage, water, sewer, and landscaping.
- **3.2.1.3** Utility design fees for permanent services.
- **3.2.2** Unless expressly stated to the contrary in the Contract Documents, the City will provide, at its own cost and expense, for the Design-Builder's information the following:
- **3.2.2.1** To the extent available, surveys describing the property, boundaries, topography and reference points for use during construction, including existing service and utility lines;
- **3.2.2.2** Temporary and permanent easements, zoning and other requirements and encumbrances affecting land use, or necessary to permit the proper design and construction of the Project and enable the Design-Builder to perform the Work;
- **3.2.2.3** A legal description of the Site;
- **3.2.2.4** To the extent available, as-built record and/or historical drawings of any existing structures at the Site;
- **3.2.2.5** To the extent available, environmental studies, reports and impact statements describing the environmental conditions, including hazardous materials, in existence at the Site; and
- **3.2.2.6** To the extent available, Geotechnical studies describing subsurface conditions, and other surveys describing other latent or concealed physical conditions at the Site.
- **3.2.3** The City will provide all City standards and guidelines, supplementary conditions and special provisions that shall be included in the plans and specifications for the Project. These may include but are not limited to: disposal of surplus material, special security provisions, investigation of underground facilities, traffic controls and regulations, special quality control testing and termite treatment requirements.
- **3.2.4** The City is responsible for securing and executing all necessary agreements with adjacent land or property owners that are necessary to enable the Design-Builder to perform the construction. The City is further responsible for all costs, including attorneys' fees, incurred in securing these necessary agreements.

3.3 Field Inspections

3.3.1 The City may utilize field inspectors to assist the City's Representative during construction in observing performance of the Design-Builder. The inspector is for the purpose of assisting the

City's Representative and should not be confused with an inspector with a City regulatory agency or with an inspector from a City laboratory pursuant to Section 2.17.

- **3.3.2** Through onsite observation of the Work in progress and field checks of materials and equipment, the inspector shall endeavor to provide protection against defects and deficiencies in the Work.
- **3.3.2.1** The inspector will be authorized to inspect all Work and materials furnished. Such inspection may extend to all or part of the Work and to the preparation, fabrication or manufacture of the materials to be used.
- **3.3.2.2** The inspector will not be authorized to issue instructions contrary to the Construction Documents or to act as foremen for the Design-Builder.
- **3.3.2.3** The inspector shall have the authority to reject Work or materials until any questions at issue can be decided by the City's Representative.
- **3.3.3** The furnishing of such services for the City shall not make the City responsible for or give the City control over construction means, methods, techniques, sequenced or procedures or for safety precautions or programs or responsibility for the Design-Builder's failure to perform the Work in accordance with Contract Documents.
- **3.4 City's Separate Contractors.** The City is responsible for all work performed on the Project or at the Site by separate contractors under the City's control. The City shall contractually require its separate contractors to cooperate with, and coordinate their activities so as not to interfere with, the Design-Builder in order to enable the Design-Builder to timely complete the Work consistent with the Contract Documents.

3.5 Intentionally Omitted

ARTICLE 4 – CONTRACT TIME

4.1 Contract Time

- **4.1.1** The Contract Time shall start with the Notice to Proceed and end with Substantial Completion.
- **4.1.2** The City shall issue a NTP letter establishing the mutually agreed upon NTP date for this Agreement and design.
- **4.1.3** The Design-Builder shall provide a Project Schedule of the design activities within seven (7) Days after the NTP.
- **4.1.3.1** The Project Schedule shall provide fourteen (14) Days to be used by the City or its designee for reviews and approvals for any interim design submissions pursuant to Section 2.8.
- **4.1.3.2** Failure on the part of the Design-Builder to adhere to the Project Schedule may be the basis for termination of this Agreement by the City.
- **4.1.4** Each GMP amendment to this Agreement will establish a separate construction NTP date, Performance Period and Substantial Completion date for the entire Project. The Performance Period(s) may be sequential or may run concurrently.
- **4.1.5** The Design-Builder agrees that it will commence performance of the Work and achieve Performance Periods and the Contract Time.
- 4.1.6 All of the times set forth in this Article 4 or by amendments to this Agreement shall be subject to

adjustment in accordance with Article 6.

4.2 Construction Schedule. Each approved GMP proposal shall include a Project Schedule as prescribed in Section 2.4 with a CPM diagram construction schedule that will indicate the path of critical activities and establish the Performance Period encompassed by the GMP. The Design-Builder will maintain the construction schedule throughout the construction.

4.3 Substantial Completion

- **4.3.1** Substantial Completion shall be for the entire Project unless a partial Substantial Completion is identified in the approved GMP schedule and stated in the Notice to Proceed letter. Substantial Completion shall be in accordance with its definition in Article 1 and with the criteria set forth in the Notice to Proceed.
- **4.3.2** Prior to notifying the City in accordance to Section 4.3.3 below, the Design-Builder shall inspect the construction and prepare and submit to the City a comprehensive list of items to be completed or corrected. The Design-Builder shall proceed promptly to complete and correct items on the list. Failure to include an item on such list does not alter the responsibility of the Design-Builder to complete all construction in accordance with the Construction Documents.
- **4.3.3** The Design-Builder shall notify the City when it believes the construction, or a portion of the construction, is substantially complete.
- **4.3.4** Within five (5) Days of the City's receipt of the Design-Builder's notice, the City and the Design-Builder will jointly inspect such construction to verify that it is substantially complete in accordance with the requirements of the Construction Documents.
- **4.3.5** If such construction is substantially complete, the City shall prepare and issue a Certificate of Substantial Completion that will set forth (i) the date of Substantial Completion of the construction or portion thereof, (ii) the remaining items of construction that have to be completed within thirty (30) Days before Final Acceptance, (iii) provisions (to the extent not already provided in the Contract Documents) establishing the City's and the Design-Builder's responsibility for the Project's security, maintenance, utilities and insurance pending Final Acceptance and (iv) an acknowledgment that warranties commence to run on the date of Substantial Completion, except as may otherwise be noted in the Certificate of Substantial Completion.
- **4.3.6** The City, at its option, may use a portion of the construction which has been determined to be substantially complete, provided, however, that (i) a Certificate of Substantial Completion has been issued for the portion of construction addressing the items set forth in Section 4.3.5 above, (ii) the Design-Builder and the City have obtained the consent of their sureties and insurers, and to the extent applicable, the appropriate government authorities having jurisdiction over the Project, and (iii) the City and the Design-Builder agree that the City's use or occupancy will not interfere with the Design-Builder's completion of the remaining construction.
- **4.4 Final Acceptance.** Upon receipt of written notice that the construction or identified portions of the construction is ready for final inspection and acceptance, the City and the Design-Builder will jointly inspect to verify that the remaining items of Work have been completed as set forth in Section 4.3.5. The City will issue a Final Acceptance letter.
- **4.5** Liquidated Damages. The Design-Builder understands that if Substantial Completion is not attained within the Contract Time as adjusted, the City will suffer damages which are difficult to determine and accurately specify. The Design-Builder agrees that if Substantial Completion is not attained within the Contract Time as adjusted, the Design-Builder shall pay the City an amount to be determined separately for each GMP as liquidated damages, and not as a penalty, for each Day that Substantial Completion extends beyond the date determined by the Contract Time as adjusted.

ARTICLE 5 – CONTRACT PRICE

5.1 Contract Price

- **5.1.1** The Contract Price will be the amounts prescribed for design in Section 5.2 below plus GMPs and Owner's Contingency.
- **5.1.2** The Contract Price is subject to adjustments made in accordance with Article 6 and by amendment.
- **5.1.2.1** GMP amendments are cumulative except for contingency. The amount of contingency for each GMP amendment will be negotiated separately.
- **5.1.2.2** If the GMP requires an adjustment due to changes in the Work, the cost of such changes is determined subject to Article 6. The markups that shall be allowed on such changes shall be no greater than the markups delineated in the approved GMP.
- **5.1.3** Unless otherwise provided in the Contract Documents, the Contract Price is deemed to include all sales, use, consumer and other taxes which are legally enacted when negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

5.2 Design and Construction Administration

- **5.2.1** The Design-Builder's fee for providing, through personnel employed by the Design-Builder or procured from qualified Subconsultants, Design Services and GMP preparation shall be a not to exceed reimbursable amount of One Million Twenty Thousand Ninety Seven and 00/100 (\$1,020,097).
- **5.2.2** The Design-Builder's fee for providing, through personnel employed by the Design-Builder or procured from qualified Subconsultants, Construction Administration services shall be a not to exceed reimbursable amount of One Hundred Ninety Seven Two Hundred Twenty and 00/100 (\$197,220).
- **5.2.3** The contract fees for the Design-Builder and Subconsultants are based upon the approved cost proposal attached hereto as Exhibit C, which is incorporated into this Agreement by reference.

5.3 GMP

- **5.3.1** At the end of the design phase or at a time determined by the City, the City will request the Design-Builder to provide a GMP or series of GMPs if the Design-Builder determines phased construction would be in the City's best interest. The approved GMP(s) will be made part of this Agreement by amendment. The GMP(s) will include amounts for completion of design, if applicable, in the same format as prescribed above and a price for construction in accordance with Section 5.3.3.
- **5.3.2** The Design-Builder guarantees to bring the completion of the design and construction within the GMP or the Design-Builder alone will be required to pay the difference between the actual cost and the GMP.
- **5.3.2.1** Unused design reimbursable and any savings of the Design-Builder's bid contingency used to buy out the construction at the conclusion of the selection of Subcontractors may be used during construction by the Design-Builder as a construction contingency.
- **5.3.2.2** Any savings realized during construction may be incorporated into the construction of the Project to fund additional scope items or will be returned to the City upon the City's request.
- 5.3.3 The GMP is composed of the following not-to-exceed cost reimbursable or lump sum amounts

defined below:

- 5.3.3.1 The Cost of the Work is actual costs and is a not-to-exceed reimbursable amount.
- **5.3.3.2** The General Conditions Costs are firm fixed lump sum amounts which will include bonds and insurance premiums based on the full contract price for construction.
- **5.3.3.3** The Construction Fee is a firm fixed lump sum.
- **5.3.3.4** The Design-Builder's Contingency is an amount the Design-Builder may use under the following conditions: (1) at its discretion for increases in the Cost of the Work or (2) with written approval of the City for increases in General Condition Costs. The Design-Builder Contingency is assumed to be a direct project cost so will receive all markups at the time of GMP submission.
- **5.3.3.5** Taxes are deemed to include all sales, use, consumer and other taxes which are legally enacted when negotiations of the GMP were concluded, whether or not yet effective or merely scheduled to go into effect. Taxes are actual costs and are a not-to-exceed reimbursable amount.
- **5.3.3.6** When the Design-Builder utilizes the Design-Builder's Contingency funds, the Design-Builder shall make the appropriate changes to the Schedule of Values with the next regular progress Payment Request. The Design-Builder shall deduct the amount of the Design-Builder's Contingency funds used from the Design-Builder's Contingency line item and add the same amount to the line item on the Schedule of Values where the funds were used. If the Design-Builder's Contingency funds are used for a new line item that was not given with the original Schedule of Values, that will be so indicated.
- **5.3.4** The Owner's Contingency are funds to be used at the sole discretion of the Owner to cover any increases in Project costs that result from Owner directed changes or unforeseen Site conditions. Owner's Contingency will be added to the GMP amount provided by the Design-Builder, the sum of which will be the full contract price for construction. Markups for Construction Fee and taxes will be applied by the Design-Builder at the time that Owner's Contingency is used.
- **5.3.5 GMPs are cumulative except for the Design-Builder's Contingency.** The amount of the Design-Builder Contingency for each GMP amendment will be negotiated separately and shall reflect the Design-Builder's risk from that point in the Project forward.

5.4 GMP Proposal

- **5.4.1** When requested, the Design-Builder shall submit three (3) sets of the approved sealed design submittal to be used to establish the GMP.
- **5.4.1.1** The Design-Builder shall sign and date the face of each document of each set used as the basis of the proposed GMP.
- **5.4.1.2** The Design-Builder shall send two (2) sets of these documents to the City's Representative, while keeping one (1) set for himself.
- **5.4.2** The GMP proposal shall include:
- **5.4.2.1** A list of the documents including the latest approved plans and design criteria, with latest issuance date including all addenda thereto which were used in preparation of the GMP proposal;
- **5.4.2.2** A list of allowances and a statement of their basis;

- **5.4.2.3** A list of the clarifications and assumptions made by the Design-Builder in the preparation of the GMP proposal, to supplement the information contained in the documents;
- **5.4.2.4** A summary of the GMP with a total for each of the components of the GMP as shown in Exhibit D attached. On the table the percentages requested should be calculated as the percent of the "Total Construction Cost." The sub-amount shown under General Conditions should be included in the General Conditions amount.
- **5.4.2.5** A spread sheet showing the basis for professional services, with hours and hourly costs for basic services and subconsultant services indicating overhead and profit.
- 5.4.2.6 A list and estimated cost of reimbursables and other not-to exceed costs.
- **5.4.2.7** A Schedule of Values for the construction, with the costs organized by subcontract categories, allowances, contingency, General Conditions costs, and the Design-Builder's construction phase fee.
- **5.4.2.8** A Project Schedule as prescribed in Section 4.2 and establishing the Performance Period and the Substantial Completion date.
- **5.4.2.9** The updated GMP as prescribed in Section 2.7.

5.5 GMP Approval

- **5.5.1** The Design-Builder shall meet with the City to review the GMP Proposal and the written statement of its basis. In the event that the City discovers any inconsistencies or inaccuracies in the information presented, the City shall promptly notify the Design-Builder, who shall make appropriate adjustments to the GMP Proposal, its basis or both.
- **5.5.2** Upon receiving the GMP proposal from the Design-Builder, the City may submit the same documents that were used by the Design-Builder in developing its GMP to an independent third party for review and verification. The third party shall develop an independent estimate and review the CPM schedule.
- **5.5.2.1** If the Design-Builder's GMP proposal is greater than the independent third party estimate, the City may require the Design-Builder to reconfirm its proposal.
- **5.5.2.2** The Design-Builder shall describe the differences between the two, and explain why its GMP reflects the scope of the Work and is correct.
- **5.5.3** If the City accepts the Design-Builder's GMP proposal and the GMP proposal is within the City's budget, the City may accept the Design-Builder's GMP proposal without comment. If the GMP Proposal exceeds the City's budget, the City must indicate in writing to the Design-Builder whether the budget has been increased to fund the excess cost.
- **5.5.4** If the City accepts the Design-Builder's GMP, the City and the Design-Builder will execute an Amendment to this Agreement for the GMP amending the Contract Price and establishing the Performance Period and the Contract Time.
- **5.5.5** The City shall authorize and cause the Design-Builder to revise the documents to the extent necessary to reflect the agreed-upon assumptions and clarifications contained in the GMP proposal.
- **5.5.6** After final submission of the GMP the City may, at the City's sole discretion and for any or no reason, accept or reject the GMP. If the City rejects the Design-Builder's GMP Proposal because it exceeds either the City's budget or the independent third party's estimate, the City may terminate the Design-Builder's Contract. If the Contract is terminated pursuant to this section, the Design-Builder's compensation shall be limited to the direct cost of its completed

Work and materials supplied as of the date of termination. Design-Builder shall not be entitled to any unearned or anticipated profit or overhead. If the Contract is terminated pursuant to this section, Design-Builder shall deliver to the City all drawings, plans, specifications, special provisions, estimates and other work entirely or partially completed, together with all unused materials supplied by the City.

ARTICLE 6 – CHANGES TO CONTRACT PRICE AND TIME

6.1 Delays

- 6.1.1 If the Design-Builder is delayed in the performance of the Work that will cause a change in the date of Substantial Completion due to acts, omissions, conditions, events, or circumstances beyond its control and due to no fault of its own or those for whom the Design-Builder is responsible, the Contract Times for performance may be reasonably extended by Change Order.
- **6.1.2** The Design-Builder shall request an increase in the Contract Time by written notice including an estimate of probable effect of delay on progress of the Work.
- **6.1.2.1** In the case of a continuing delay only one (1) request is necessary.
- **6.1.2.2** Written notice shall be received within thirty (30) Days of the commencement of the cause.
- **6.1.2.3** Written notice received more than thirty (30) days after commencement of the cause, the period of delay shall be deemed to commence thirty (30) days prior to the giving of such notice.
- **6.1.3** By way of example, events that will entitle the Design-Builder to an extension of the Contract Time include acts or omissions of the City or anyone under the City's control (including separate contractors), changes in the Work, Differing Site Conditions, Hazardous Conditions, delays by regulating agencies, wars, floods, labor disputes, unusual delay in transportation, epidemics abroad, earthquakes, adverse weather conditions not reasonably anticipated, and other acts of God.
- **6.1.4** If adverse weather conditions are the basis for a request for additional Contract Time, such requests shall be documented by data substantiating that weather conditions were abnormal for the period of time and could not have been reasonably anticipated, and that the weather had an adverse effect on the scheduled Substantial Completion date.
- **6.1.5** It is understood, however, that permitting the Design-Builder to proceed to complete any Work, or any part of them, after the date to which the time of completion may have been extended, shall in no way act as a waiver on the part of the City of any of its legal rights herein.
- **6.1.6** In addition to the Design-Builder's right to a time extension for those events set forth in this Section, the Design-Builder may also be entitled to an appropriate adjustment of the Contract Price provided, however, that the Contract Price shall not be adjusted for those events set forth in this Section that are beyond the control of both the Design-Builder and the City, including the events of war, floods, labor disputes, earthquakes, epidemics, adverse weather conditions not reasonably anticipated, and other acts of God.

6.2 Differing Site Conditions

- **6.2.1** If the Design-Builder encounters a Differing Site Condition, the Design-Builder may be entitled to a Change Order to adjust the Contract Price and/or Contract Time(s) to the extent the Design-Builder's cost and/or time of performance are impacted by the Differing Site Condition.
- **6.2.2** Upon encountering a Differing Site Condition, the Design-Builder shall provide prompt written notice to the City of such condition, which notice shall not be later than seven (7) Days after such condition has been encountered. The Design-Builder shall, to the extent reasonably possible, provide such notice before the Differing Site Condition has been substantially

disturbed or altered.

6.3 Legal Requirements. The Contract Price and/or Contract Time(s) shall be adjusted by Change Order to compensate the Design-Builder for the effects of any changes in the Legal Requirements enacted after the date of this Agreement affecting the performance of the Work, or if a Guaranteed Maximum Price is established after the date of this Agreement, the date the parties agree upon the Guaranteed Maximum Price. Such effects may include, without limitation, revisions the Design-Builder is required to make to the Construction Documents because of changes in Legal Requirements.

6.4 City Requested Change in Work

- **6.4.1** The City reserves the right to make, at any time during the progress of the Work such alterations as may be found necessary or desirable.
- **6.4.2** Such alterations and changes shall not invalidate this Agreement and the Design-Builder agrees to perform the services as altered, the same as if it has been a part of the original Contract.
- **6.4.3** Upon receipt of a request for proposal for a change in Work, the Design-Builder shall prepare a proposal in significant detail, using the rates and markups established in the Contract Documents as a basis of the Contract Price adjustment. The Design-Builder's proposal shall include a detailed description of any schedule impact.
- **6.4.4** The City and the Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments. Upon reaching an agreement, the parties shall prepare and execute an appropriate Change Order reflecting the terms of the adjustment.
- **6.4.5** If the City requests a proposal for a change in the Work from the Design-Builder and subsequently elects not to proceed with the change, a Change Order may be issued to reimburse the Design-Builder for reasonable costs incurred for estimating services and other services involved in the preparation of proposed revisions to the Contract Documents.
- 6.5 Minor Changes. The City may make minor changes in the Work that do not involve an adjustment in the Contract Price and/or Contract Time(s) and do not materially and adversely affect the Work, including the design, quality, performance and workmanship required by the Contract Documents. The Design-Builder may make minor changes in the Work consistent with the intent of the Contract Documents, provided, however that the Design-Builder shall promptly inform the City, in writing, of any such changes and record such changes on the documents maintained by the Design-Builder.
- 6.6 Emergencies. In any emergency affecting the safety of persons and/or property, the Design-Builder shall act, at its discretion, to prevent threatened damage, injury or loss. Any change in the Contract Price and/or Contract Time(s) on account of emergency work shall be determined as provided in this Article 6.

ARTICLE 7- PROCEDURE FOR PAYMENT

7.1 Request for Payment for Design Phase Services

- **7.1.1** The Design-Builder will be paid one hundred percent (100%) of the amount earned for design services provided and in accordance with the retention requirements identified in Section 7.4 below.
- **7.1.2** After approval of Construction Documents in accordance with Section 2.8 and GMP(s) for the total construction of the Project in accordance with section 5.4, the Design-Builder will be paid

one hundred percent (100%) of the amount for design services less the total of payments previously made.

- **7.1.3** The Design-Builder shall pay all sums due to Subconsultants for services and reimbursable expenses within fourteen (14) Days after the Design-Builder has received payment for those services from the City.
- **7.1.4** Requests for monthly payments by the Design-Builder shall be submitted on the "Design Services Payment Request" form and shall be accompanied by a design progress report, detailed invoices and receipts, if applicable. This submittal shall include, as a minimum, a narrative description of the tasks accomplished during the billing period, a listing of any deliverables submitted, and the Subconsultants' actual requests for payment plus similar narrative and listing of their work.
- **7.1.5** Payments for those services negotiated as a lump sum shall be made in accordance with the percentage of the services completed during the preceding month. Those services negotiated as a not-to-exceed fee shall be paid in accordance with the actual cost of the service expended during the preceding month.
- 7.1.6 All Payment Requests shall be submitted to the City for review and approval.
- **7.2 Requests for Payment for Construction Services.** The Design-Builder agrees at its own proper cost and expense, to do all construction, as called for by this Agreement as amended free and clear of all claims, liens, and charges whatsoever, in the manner and under the conditions specified within the time, or times, stated in this Agreement as amended.
- **7.2.1** The Schedule of Values submitted will serve as the basis for monthly progress payments made to the Design-Builder throughout the construction.
- **7.2.2** The Design-Builder shall submit to the City on the monthly anniversary of the construction NTP date beginning with the first month after the construction NTP date the "Construction Payment Request " form.
- **7.2.3** At least five (5) working days prior to the date established for a Construction Payment Request, the Design-Builder shall submit an updated Project Schedule and meet with the City's Representative to review the progress of the construction, as it will be reflected on the Construction Payment Request.
- **7.2.4** The Design-Builder Construction Payment Request may request payment for equipment and materials not yet incorporated into the Project if construction progress is in reasonable conformance with the approved schedule.
- **7.2.4.1** For equipment and materials suitably stored at the Site, the equipment and materials shall be protected by suitable insurance and the City shall receive the equipment and materials free and clear of all liens and encumbrances.
- **7.2.4.2** For materials and equipment stored off the Site, the City must approve the storage facility. The material and equipment must be stored within Maricopa or Pinal County and be accessible for Owner's inspection. Title to such materials and equipment protects the City's interest and shall include applicable insurance, bonding, storage and transportation to the Site.
- **7.2.4.3** All bonds and insurance required for stored materials shall name the City as the loss payee to the extent of its interest in the stored materials.
- **7.2.5** The Construction Payment Request shall constitute the Design-Builder's representation that the construction has been performed consistent with the Contract Documents, has progressed to the point indicated in the Construction Payment Request, and will pass to the City free and clear of all claims, liens, encumbrances, and security interests upon the incorporation of the construction into the Project.

7.3 Progress Payment

- **7.3.1** Requests for monthly payments by the Design-Builder for design phase services shall be submitted on the City's "Payment Request" form and shall be accompanied by a design progress report, detailed invoices and receipts, if applicable. This submittal shall include, as a minimum, a narrative description of the tasks accomplished during the billing period, a listing of any deliverables submitted, and the subconsultants' actual requests for payment plus similar narrative and listing of their work.
- **7.3.2** The City shall make payment for construction phase services in accordance with A.R.S. §34-609 such that payment will be made no later than fourteen (14) days after the work is certified and approved. The City shall review payment request and make recommendation of approval or denial within seven (7) calendar days after the City's receipt of each properly submitted and accurate Construction Payment Request, but in each case less the total of payments previously made, and less amounts properly withheld under Section 7.4.2 below.
- **7.3.3** The City shall pay the Design-Builder all amounts properly due. If the City determines that the Design-Builder is not entitled to all or part of the Construction Payment Request, it will notify the Design-Builder in writing at least seven (7) days after the date the Construction Payment Request is received by the City. The notice shall indicate the specific amounts the City intends to withhold, the reasons and contractual basis for the withholding, and the specific measures the Design-Builder must take to rectify the City's concerns. The Design-Builder and the City will attempt to resolve the City's concerns. If the parties cannot resolve such concerns, the Design-Builder may pursue its rights under the Contract Documents, including those under Article 8 hereof.
- **7.3.4** Notwithstanding anything to the contrary in the Contract Documents, the City shall pay the Design-Builder all undisputed amounts in a Payment Request within the times required by the Agreement.

7.4 Retention on Progress Payments

7.4.1 Design Services

- **7.4.1.1** In no event shall the City pay more than ninety percent (90%) of the total amount for design services until acceptance of all design Work.
- **7.4.1.2** After approval of Construction Documents in accordance with Section 2.8.8, the Design-Builder will be paid one hundred percent (100%) of the amount for design services less the total of payments previously made.
- **7.4.1.3** The Design-Builder shall pay all sums due to Subconsultants for services and reimbursable expenses within fourteen (14) Days after the Design-Builder has received payment for those services from the City.

7.4.2 Construction Services

7.4.2.1 The City will retain ten percent (10%) of the amount for construction services on each Construction Payment Request provided, however, that when fifty percent (50%) of the construction has been completed by the Design-Builder, upon request of the Design-Builder, the City may reduce the amount retained to five percent (5%) from the Design-Builder's subsequent Construction Payment Request if the Design-Builder's performance of work has been satisfactory.

7.5 Substantial Completion

7.5.1 Upon Substantial Completion of the entire Work or, if applicable, any portion of the Work, the City shall release to the Design-Builder all retained amounts relating, as applicable, to the entire Work or completed portion of the Work, less an amount up to two and one half times (2.5) the reasonable value of all remaining or incomplete items of Work as noted in the Certificate of Substantial Completion.

7.6 Final Payment

- **7.6.1** After receipt of a final Payment Request from the Design-Builder, the City shall make final payment sixty (60) days after the receipt by the City, provided that a Final Acceptance Letter has been issued by the City in accordance with Section 4.4.
- **7.6.2** At the time of submission of its final Payment Request, the Design-Builder shall provide the following information:
- **7.6.2.1** An affidavit that there are no claims, obligations or liens outstanding or unsatisfied for labor, services, material, equipment, taxes or other items performed, furnished or incurred for or in connection with the Work which will in any way affect the City's interests;
- **7.6.2.2** A general release executed by the Design-Builder waiving, upon receipt of final payment by the Design-Builder, all claims, except those claims previously made in writing to the City and remaining unsettled at the time of final payment; and
- 7.6.2.3 Consent of the Design-Builder's surety, if any, to final payment.

7.7 Payments To Subcontractors or Supplier

- 7.7.1 The Design-Builder shall pay its Subcontractors or suppliers within seven (7) Days of receipt of each progress payment from the City. The Design-Builder shall pay for the amount of Work performed or materials supplied by each Subcontractor or supplier as accepted and approved by the City with each progress payment. In addition, any reduction of retention by the City to the Design-Builder shall result in a corresponding reduction to Subcontractors or suppliers who have performed satisfactory work. The Design-Builder shall pay Subcontractors or suppliers the reduced retention within fourteen (14) Days of the payment of the reduction of the retention to the Design-Builder. No Contract between the Design-Builder and its Subcontractors and suppliers may materially alter the rights of any Subcontractor or supplier to receive prompt payment and retention reduction as provided herein.
- **7.7.2** If the Design-Builder fails to make payments in accordance with these provisions, the City may take any one or more of the following actions and the Design-Builder agrees that the City may take such actions:
- 7.7.2.1 To hold the Design-Builder in default under this Agreement;
- **7.7.2.2** Withhold future payments including retention until proper payment has been made to Subcontractors or suppliers in accordance with these provisions;
- **7.7.2.3** Reject all future offers to perform work for the City from the Design-Builder for a period not to exceed one (1) year from Substantial Completion date of this Project;
- 7.7.2.4 Terminate this Agreement.
- **7.7.3** If the Design-Builder's payment to a Subcontractor or supplier is in dispute, the Design-Builder and Subcontractor or supplier agree to submit the dispute to any of one of the following dispute resolution processes within fourteen (14) Days from the date any party gives notice to the others: (i) binding arbitration; (ii) a form of alternative dispute resolution (ADR) agreeable to all

parties or (iii) a City of Maricopa facilitated mediation. When disputed claim is resolved through ADR or otherwise, the Design-Builder and Subcontractor or supplier agree to implement the resolution within seven (7) Days from the resolution date.

- **7.7.4** Should the City fail or delay in exercising or enforcing any right, power, privilege, or remedy under this Section, such failure or delay shall not be deemed a waiver, release, or modification of the requirements of this Section or of any of the terms or provisions thereof.
- **7.7.5** The Design-Builder shall include these prompt payment provisions in every subcontract, including procurement of materials and leases of equipment for this Agreement.

7.8 Record Keeping and Finance Controls

- **7.8.1** Records of the Design-Builder's direct personnel payroll, reimbursable expenses pertaining to this Project and records of accounts between the City and the Design-Builder shall be kept on a generally recognized accounting basis and shall be available for up to three (3) years following final completion of the Project.
- **7.8.2** The City, its authorized representative, and/or the appropriate federal agency, reserve the right to audit the Design-Builder's records to verify the accuracy and appropriateness of all pricing data, including data used to negotiate Contract Documents and any Change Orders.
- **7.8.3** The City reserves the right to decrease the Contract Price and/or payments made on this Agreement if, upon audit of the Design-Builder's records, the audit discloses the Design-Builder has provided false, misleading, or inaccurate cost and pricing data.
- **7.8.4** The Design-Builder shall include a similar provision in all of its agreements with Subconsultants and Subcontractors providing services under the Contract Documents to ensure the City, its authorized representative, and/or the appropriate federal agency, has access to the Subconsultants' and Subcontractors' records to verify the accuracy of cost and pricing data.
- **7.8.5** The City reserves the right to decrease the Contract Price and/or payments made on this Agreement if the above provision is not included in Subconsultant's and Subcontractor's contracts, and one or more Subconsultants and/or Subcontractors do not allow the City to audit their records to verify the accuracy and appropriateness of pricing data.
- **7.9** Nothing contained in this Section shall provide a basis for any Subcontractor to assert any claim against the City for its administration, enforcement or waiver of the provisions of this prompt payment provision.

ARTICLE 8- CLAIMS AND DISPUTES

8.1 Requests for Contract Adjustments and Relief

- **8.1.1** If either the Design-Builder or the City believes that it is entitled to relief against the other for any event arising out of or related to the Work, such party shall provide written notice to the other party of the basis for its claim for relief.
- **8.1.2** Such notice shall, if possible, be made prior to incurring any cost or expense and in accordance with any specific notice requirements contained in applicable sections of the Agreement.
- **8.1.3** In the absence of any specific notice requirement, written notice shall be given within a reasonable time, not to exceed twenty-one (21) Days, after the occurrence giving rise to the claim for relief or after the claiming party reasonably should have recognized the event or condition giving rise to the request, whichever is later.

8.1.4 Such notice shall include sufficient information to advise the other party of the circumstances giving rise to the claim for relief, the specific contractual adjustment or relief requested and the basis of such request.

8.2 Dispute Avoidance and Resolution

- **8.2.1** The parties are fully committed to working with each other throughout the Project and agree to communicate regularly with each other at all times so as to avoid or minimize disputes or disagreements. If disputes or disagreements do arise, the Design-Builder and the City each commit to resolving such disputes or disagreements in an amicable, professional and expeditious manner so as to avoid unnecessary losses, delays and disruptions to the Work.
- **8.2.2** The Design-Builder and the City will first attempt to resolve disputes or disagreements at the field level through discussions between the Design-Builder's Representative and the City's Representative.
- 8.2.3 If a dispute or disagreement cannot be resolved through the Design-Builder's Representative and the City's Representative, the Design-Builder's Senior Representative and the City's Senior Representative, upon the request of either party, shall meet as soon as conveniently possible, but in no case later than thirty (30) Days after such a request is made, to attempt to resolve such dispute or disagreement. Prior to any meetings between the Senior Representatives, the parties will exchange relevant information that will assist the parties in resolving their dispute or disagreement.
- **8.3 Duty to Continue Performance.** Unless provided to the contrary in the Contract Documents, the Design-Builder shall continue to perform the Work and the City shall continue to satisfy its payment obligations to the Design-Builder pending the final resolution of any dispute or disagreement between the Design-Builder and the City.

8.4 Representatives of the Parties

8.4.1 <u>City's Representatives</u>

8.4.1.1 The City designates the individual listed below as its Senior Representative ("City's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 8.2.3:

Brent Billingsley, Development Services Director City of Maricopa, Development Services Department 45145 W. Madison Ave Maricopa, Arizona 85139

8.4.1.2 The City designates the individual listed below as its City's Representative, which individual has the authority and responsibility set forth in Section 8.2.2:

Adam Brill, City's Program Manager ABACUS Project Management 3030 N. Central Ave, Suite 1207 Phoenix, AZ 85012 (602)282-8860 Phone

- 8.4.2 Design-Builder's Representatives
- **8.4.2.1** The Design-Builder designates the individual listed below as its Senior Representative ("Design-Builder's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 8.2.3:

Fritz Behrhorst, Vice President of Pre-Construction <u>FBehrhorst@Haydonbc.com</u> Or Les Keeble, Vice President of Operations <u>LKeeble@Haydonbc.com</u> Haydon Building Corp 4640 E Cotton Gin Loop Phoenix, AZ 85040 602-296-1496

8.4.2.2 The Design-Builder designates the individual listed below as its Design-Builder's Representative, which individual has the authority and responsibility set forth in Section 8.2.2:

K.T. "Cub" Carter, Senior Project Manager <u>KCarter@Haydonbc.com</u> Haydon Building Corp 4640 E Cotton Gin Loop Phoenix, AZ 85040 602-296-1496

ARTICLE 9 – SUSPENSION AND TERMINATION

9.1 City's Right to Stop Work

- **9.1.1** The City may, at its discretion and without cause, order the Design-Builder in writing to stop and suspend the Work. Immediately after receiving such notice, the Design-Builder shall discontinue advancing the Work specified under this Agreement.
- **9.1.2** Such suspension shall not exceed one hundred and eighty (180) consecutive Days during the duration of the Project.
- **9.1.3** The Design-Builder may seek an adjustment of the Contract Price and Time, if its cost or time to perform the Work has been adversely impacted by any suspension or stoppage of Work by the City.
- **9.2** Termination for Convenience. Upon receipt of written notice to the Design-Builder, the City may, at its discretion and without cause, elect to terminate this Agreement. If the City suspends the Work for one hundred and eighty one (181) consecutive Days or more, such suspension shall be deemed a termination for convenience.
- **9.2.1** Upon such termination during Design Services, the Design-Builder shall deliver to the City all drawings, plans, specifications, special provisions, estimates and other work entirely or partially completed, together with all unused materials supplied by the City.
- **9.2.1.1** The Design-Builder shall estimate the value of the Work it has completed and submit its appraisal to the City for evaluation. The City shall have the right to inspect the Subconsultant's Work to appraise the Work completed.
- **9.2.1.2** The Design-Builder shall receive compensation for services performed to the date of such termination as set forth in Section 9.2.4. The fee shall be paid in accordance with Article 7 of this Agreement, and shall be an amount mutually agreed upon by the Design-Builder and the City. If there is no mutual agreement, the final determination shall be made in accordance with Article 8.
- 9.2.1.3 The Design-Builder shall not be entitled to anticipated profit or anticipated overhead. In no event

shall the fee exceed that set forth in Article 5 of this Agreement or as amended.

- **9.2.1.4** The City shall make the final payment within sixty (60) Days after the Design-Builder has delivered the last of the partially completed items and the final fee has been agreed upon.
- **9.2.1.5** If the City terminates this Agreement pursuant to this Section and proceeds to design and construct the Project through its employees, agents or third parties, the City's rights to use the work product shall be as set forth in Section 12.4 hereof.
- **9.2.2** Upon such termination during construction services, the Design-Builder shall proceed with the following obligations:
- 9.2.2.1 Stop Work as specified in the notice.
- 9.2.2.2 Place no further subcontracts or orders.
- **9.2.2.3** Terminate all subcontracts to the extent they relate to the Work terminated.
- **9.2.2.4** Assign to the City all right, title and interest of the Design-Builder under the subcontracts terminated, in which case the City shall have the right to settle or to pay any termination settlement proposal arising out of those terminations.
- **9.2.2.5** Take any action that may be necessary for the protection and preservation of the property related to the Contract that is in the possession of the Design-Builder and which the City has or may acquire an interest.
- **9.2.3** The Design-Builder shall submit complete termination inventory schedules no later than sixty (60) Days from the date of the notice of termination.
- **9.2.4** The City shall pay the Design-Builder the following:
- **9.2.4.1** The direct value of its completed Work and materials supplied as of the date of termination.
- **9.2.4.2** The reasonable costs and expenses attributable to such termination.
- **9.2.4.3** The Design-Builder shall be entitled to profit and overhead on completed Work only, but shall not be entitled to anticipated profit or anticipated overhead. If it appears the Design-Builder would have sustained a loss on the entire Work had the Project been completed, the Design-Builder shall not be allowed profit and the City shall reduce the settlement to reflect the indicated rate of loss.
- **9.2.5** The Design-Builder shall maintain all records and documents for three (3) years after final settlement. These records shall be maintained and subject to auditing as prescribed in Section 7.8.

9.3 City's Right to Perform and Terminate for Cause

- **9.3.1** If the City provides the Design-Builder with a written order to correct deficiencies to provide adequate maintenance of traffic, adequate cleanup, adequate dust control, or to repair damage resulting from abnormal weather conditions, and the Design-Builder fails to comply in a time frame specified, the City may have the deficiencies corrected by other sources at the Design-Builder's expense.
- 9.3.2 If the Design-Builder persistently fails to (i) provide a sufficient number of skilled workers, (ii) supply the materials required by the Contract Documents, (iii) comply with applicable Legal Requirements, (iv) timely pay, without cause, Subconsultants and/or Subcontractors, (v) prosecute the Work with promptness and diligence to ensure that the Work is completed by the Contract Time, as such times may be adjusted, or (vi) perform material obligations under the Contract Documents, then the City, in addition to any other rights and remedies provided in the

Contract Documents or by law, shall have the rights set forth in Sections 9.3.3 and 9.3.4 below.

- **9.3.3** Upon the occurrence of an event set forth in Section 9.3.2 above, the City may provide written notice to the Design-Builder that it intends to terminate the Agreement unless the problem cited is cured, or commenced to be cured, within seven (7) Days of the Design-Builder's receipt of such notice.
- **9.3.3.1** If the Design-Builder fails to cure, or reasonably commence to cure, such problem, then the City may give a second written notice to the Design-Builder of its intent to terminate within an additional seven (7) Day period.
- **9.3.3.2** If the Design-Builder, within such second seven (7) Day period, fails to cure, or reasonably commence to cure, such problem, then the City may declare the Agreement terminated for default by providing written notice to the Design-Builder of such declaration.
- **9.3.4** Upon declaring the Agreement terminated pursuant to Section 9.3.3.2 above, the City may enter upon the premises and take possession, for the purpose of completing the Work, of all materials, equipment, scaffolds, tools, appliances and other items thereon, which have been purchased or provided for the performance of the Work, all of which the Design-Builder hereby transfers, assigns and sets over to the City for such purpose, and to employ any person or persons to complete the Work and provide all of the required labor, services, materials, equipment and other items.
- **9.3.5** In the event of such termination, the Design-Builder shall not be entitled to receive any further payments under the Contract Documents until the Work shall be finally completed in accordance with the Contract Documents. At such time, the Design-Builder will only be entitled to be paid for Work performed and accepted by the City prior to its default.
- **9.3.6** If the City's cost and expense of completing the Work exceeds the unpaid balance of the Contract Price, then the Design-Builder shall be obligated to pay the difference to the City. Such costs and expense shall include not only the cost of completing the Work, but also losses, damages, costs and expense, including attorneys' fees and expenses, incurred by the City in connection with the re-procurement and defense of claims arising from the Design-Builder's default.
- **9.3.7** If the City improperly terminates the Agreement for cause, the termination for cause shall be converted to a termination for convenience in accordance with the provisions of Section 9.2.

ARTICLE 10 – INSURANCE AND BONDS

10.1 Insurance Requirements

- **10.1.1** Concurrently with the execution of the Agreement, the Design-Builder shall furnish the City of Maricopa a certificate of insurance on a standard insurance industry ACORD form. The ACORD form shall be issued by an insurance company authorized to transact business in the State of Arizona, or one that is named on the List of Qualified Unauthorized Insurers maintained by the Arizona Department of Insurance.
- **10.1.2** The Design-Builder, Subcontractors and subconsultants shall procure and maintain until all of their obligations have been discharged, including any warranty periods under this Agreement are satisfied, insurance against claims for injury to persons or damage to property, which may arise from or in connection with the performance of the Work hereunder by the Design-Builder, his agents, representatives, employees, or Subcontractors.
- **10.1.3** The insurance requirements herein are minimum requirements for this Agreement and in no way limit the indemnity covenants contained in this Agreement.

- **10.1.4** The City in no way warrants that the minimum limits contained herein are sufficient to protect the Design-Builder from liabilities that might arise out of the performance of the Work under this Agreement by the Design-Builder, his agents, representatives, employees, Subcontractors or subconsultants and the Design-Builder is free to purchase such additional insurance as may be determined necessary.
- **10.2 Minimum Scope And Limits Of Insurance.** The Design-Builder shall provide coverage at least as broad and with limits of liability not less than those stated below. An excess liability policy or umbrella liability policy may be used to meet the minimum liability requirements provided that the coverage is written on a "following form" basis.

10.2.1 Commercial General Liability-Occurrence Form

Policy shall include bodily injury, property damage, broad form contractual liability and XCU coverage.

1. For Projects with an estimated Project-Construction value of Less than \$10,000,000:

General Aggregate	\$2,000,000
Products-Completed Operations Aggregate	\$1,000,000
Personal & Advertising Injury	\$1,000,000
Each Occurrence	\$1,000,000

2. For Projects with an estimated Project-Construction value of \$10,000,000 to \$25,000,000:

General Aggregate	\$5,000,000
Products-Completed Operations Aggregate	\$5,000,000
Personal & Advertising Injury	\$5,000,000
Each Occurrence	\$5,000,000

3. For Projects with an estimated Project-Construction value of \$25,000,000 to \$50,000,000:

General Aggregate	\$10,000,000
Products-Completed Operations Aggregate	\$10,000,000
Personal & Advertising Injury	\$10,000,000
Each Occurrence	\$10,000,000

4. For Project with an Estimated Project-Construction value over \$50,000,000. Contact Risk Management for determination of Limits and type of insurance program.

The policy shall be endorsed to include the following additional insured language: "The City of Maricopa shall be named as an additional insured with respect to liability arising out of the activities performed by, or on behalf of the Design-Builder, including completed operations."

10.2.2 Automobile Liability-

Bodily injury and property damage for any owned, hired, and non-owned vehicles used in the performance of this Agreement

Combined Single Limit (CSL)

\$1,000,000

The policy shall be endorsed to include the following additional insured language: "The City of Maricopa shall be named as an additional insured with respect to liability arising out of the activities performed by, or on behalf of the Design-Builder including automobiles owned, leased, hired or borrowed by the Design-Builder".

10.2.3 Workers Compensation and Employers Liability

Workers Compensation

Employers' Liability	
Each Accident	\$ 100,000
Disease-Each Employee	\$ 100,000
Disease-Policy Limit	\$ 500,000

The policy shall contain a waiver of subrogation against the City of Maricopa.

This requirement shall not apply when a Design-Builder or subconsultant is exempt under A.R.S. §23-901, AND when such Design-Builder or subconsultant executed the appropriate sole proprietor waiver form.

10.2.4 Professional Liability

This policy shall cover professional misconduct or lack of ordinary skill for those positions defined in the Scope of Services of this Contract.

Each Claim	\$1,000,000
Annual Aggregate	\$2,000,000

- **10.2.4.1** The required professional liability coverage must cover work done or to be done or on the behalf of the Design-Builder.
- **10.2.4.2** In the event that professional liability insurance required by this Agreement is written on a "claims made" basis, coverage shall be maintained for two (2) years past completion and acceptance of the Work or services required by this Contract.
- **10.2.4.3** Any professional liability shall specifically delete any design-build or similar exclusions that could compromise coverage's because of the design-build delivery of the Project.

10.2.5 Builders' Risk Insurance or Installation Floater

In an amount equal to the initial Contract Price plus additional coverage equal to contract amount for all subsequent change orders. (Submitted at the time of construction)

- **10.2.5.1** The City of Maricopa, the Design-Builder, Subcontractors, design professional and design professional's consultant and any others with an insurable interest in the Work shall be Named Insureds on the policy.
- **10.2.5.2** Coverage shall be written on an all risk, replacement cost basis and shall include coverage for soft costs, flood and earth movement.
- **10.2.5.3** Policy shall be maintained until whichever of the following shall first occur: (i) final payment has been made; or, (ii) until no person or entity, other than the City of Maricopa, has an insurable interest in the property required to be covered.
- **10.2.5.4** Policy shall be endorsed such that the insurance shall not be canceled or lapse because of any partial use or occupancy by the City.
- **10.2.5.5** Policy must provide coverage from the time any covered property becomes the responsibility of the Design-Builder, and continue without interruption during construction, renovation, or installation, including any time during which the covered property is being transported to the construction installation site, or awaiting installation, whether on or off Site.
- **10.2.5.6** Policy shall contain a waiver of subrogation against the City of Maricopa.
- **10.2.5.7** The Design-Builder is responsible for the payment of all policy deductibles.

- **10.3** Additional Insurance Requirements. The policies shall include, or be endorsed to include, the following provisions:
- **10.3.1** On insurance policies where the City of Maricopa is named as an additional insured, the City of Maricopa shall be an additional insured to the full limits of liability purchased by the Design-Builder even if those limits of liability are in excess of those required by this Agreement.
- **10.3.2** The Design-Builder's insurance coverage shall be primary insurance and non-contributory with respect to all other available sources.
- **10.3.3** Coverage provided by the Design-Builder shall not be limited to the liability assumed under the indemnification provisions of this Agreement.
- **10.4 Subconsultant's and Subcontractor's Insurance.** The Design-Builder's certificate(s) shall include all subcontractors as additional insureds under its policies or Subcontractors shall maintain separate insurance as determined by the Design-Builder, however, subcontractor's limits of liability shall not be less than \$1,000,000 per occurrence / \$2,000,000 aggregate. All coverage's for Subcontractors and Subconsultants shall be appropriate to cover all of its Work performed herein.

10.5 Notice Of Cancellation

For each insurance policy required by the insurance provisions of this Contract, the Design-Builder must provide to the City, within two (2) business days of receipt, a notice if a policy is suspended, voided or cancelled for any reason. Such notice shall be sent directly to the City's named Project Manager at:

City of Maricopa Development Services Department 45145 W. Madison Ave Maricopa, Arizona 85139

10.6 Acceptability Of Insurers

Insurance is to be placed with insurers duly licensed or authorized to do business in the State of Arizona and with an A. M. Best's rating of no less than B+ VI. The City in no way warrants that the above required minimum insurer rating is sufficient to protect the Design-Builder from potential insurer insolvency.

10.7 Verification of Coverage

- **10.7.1** The Design-Builder shall furnish the City Certificates of Insurance (ACORD form or equivalent approved by the City) and with original endorsements effecting coverage as required by this Agreement. The certificates and endorsements for each insurance policy are to be signed by a person authorized by that insurer to bind coverage on its behalf. *Any policy endorsements that restrict or limit coverages shall be clearly noted on the certificate of insurance.*
- **10.7.2** All certificates and any required endorsements are to be received and approved by the City before Work commences. Each insurance policy required by this Agreement must be in effect at or prior to the earlier of commencement of Work under this Agreement or the signing of this Agreement and remain in effect for the duration of the Project. Failure to maintain the insurance policies as required by this Agreement or to provide evidence of renewal is a material breach of contract.
- **10.7.3** All certificates of insurance required by this Agreement shall be sent directly to the City of Maricopa. The project description shall be included on the Certificates of Insurance. The City reserves the right to require complete, certified copies of all insurance policies required by this Agreement, at any time.

- **10.8 Approval.** Any modification or variation from the insurance requirements in this Agreement shall be approved by the City Attorney's Office, whose decision shall be final. Such action will not require a formal contract amendment, but may be made by administrative action.
- **10.9 Bonds and Other Performance Security.** The Design-Builder shall provide the following performance bond and labor and material payment bond:
- **10.9.1** Prior to execution of this Agreement, the Design-Builder must provide a performance bond and a labor and materials bond, each in an amount equal to the amount of initial Contract Price designated for construction services set forth in Article 5.
- **10.9.2** Each such bond shall be executed by a surety company or companies holding a Certificate of Authority to transact surety business in the State of Arizona, issued by the Director of the Arizona Department of Insurance. A copy of the Certificate of Authority shall accompany the bonds. The Certificate shall have been issued or updated within two (2) years prior to the execution of the Agreement.
- **10.9.3** The bonds shall be made payable and acceptable to the City of Maricopa.
- **10.9.4** The bonds shall be written or countersigned by an authorized representative of the surety who is either a resident of the State of Arizona or whose principal office is maintained in this state, as by law required, and the bonds shall have attached thereto a certified copy of Power of Attorney of the signing official. If one Power of Attorney is submitted, it shall be for twice the total Contract amount. If two Powers of Attorney are submitted, each shall be for the total Contract amount. Personal or individual bonds are not acceptable.
- **10.9.5** Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Agreement, the Design-Builder shall promptly furnish a copy of the bonds or shall permit a copy to be made.
- **10.9.6** All bonds submitted for this project shall be provided by a company which has been rated "A- or better" by the A.M. Best Company.

ARTICLE 11 - INDEMNIFICATION

11.1 Intellectual Property

- **11.1.1** The Design-Builder shall pay all royalties and license fees associated with its performance of services herewith.
- **11.1.2** The Design-Builder shall defend any action or proceeding brought against the City based on any claim that the Work, or any part thereof, or the operation or use of the Work or any part thereof, constitutes infringement of any United States patent or copyright, now or hereafter issued. The City shall give prompt written notice to the Design-Builder of any such action or proceeding and will reasonably provide authority, information and assistance in the defense of same. The Design-Builder shall indemnify and hold harmless the City from and against all damages and costs, including but not limited to attorneys' fees and expenses awarded against the City or the Design-Builder in any such action or proceeding. The Design-Builder agrees to keep the City informed of all developments in the defense of such actions.
- **11.1.3** If the City is enjoined from the operation or use of the Work, or any part thereof, as the result of any patent or copyright suit, claim, or proceeding, the Design-Builder shall at its sole expense take reasonable steps to procure the right to operate or use the Work. If the Design-Builder cannot so procure such right within a reasonable time, the Design-Builder shall promptly, at the Design-Builder's option and at the Design-Builder's expense, (i) modify the Work so as to avoid infringement of any such patent or copyright or (ii) replace said Work with Work that does not infringe or violate any such patent or copyright.

- **11.1.4** Sections 11.1.2 and 11.1.3 above shall not be applicable to any suit, claim or proceeding based on infringement or violation of a patent or copyright (i) relating solely to a particular process or product of a particular manufacturer specified by the City and not offered or recommended by the Design-Builder to the City or (ii) arising from modifications to the Work by the City or its agents after acceptance of the Work.
- **11.1.5** The obligations set forth in this Section 11.1 shall constitute the sole agreement between the parties relating to liability for infringement of violation of any patent or copyright.

11.2 General Indemnification

- **11.2.1** The Design-Builder hereby agrees to defend, indemnify and hold harmless the City, any of its departments, agencies, officers, or employees from all damages, claims or liabilities and expenses (including reasonable Attorney's fees) arising out of or resulting in any way from the performance of professional services for the City in the Design-Builder's capacity as a Design-Builder, and caused by any willful or negligent error, omission, or act of the Design-Builder or any person employed by it or anyone for whose acts the Design-Builder is legally liable. In consideration of the award of this contract, the Design-Builder agrees to waive all rights of subrogation against the City, its officers, agents and employees for losses arising from the work performed by the Design-Builder for the City.
- **11.2.2** The Design-Builder agrees to indemnify and save harmless the City of Maricopa, its officers, agents and employees, and any jurisdiction or agency issuing permits for any Work included in the Project, their officers, agents and employees, hereinafter referred to as indemnitee, from all suits and claims, including attorney's fees and cost of litigation, actions, loss, damage, expense, cost or claims of any character or any nature arising out of the Work done in fulfillment of the terms of this Agreement or on account of any act, claim or amount arising or recovered under workmen's compensation law or arising out of the failure of the Design-Builder to conform to any statutes, ordinances, regulation, law or court decree. It is agreed that the Design-Builder will be responsible for primary loss investigation, defense and judgment costs where this Agreement of indemnity applies. In consideration of the award of this Agreement, the Design-Builder agrees to waive all rights of subrogation against the City, its officers, officials, agents and employees for losses arising from the work performed by the Design-Builder for the City.

ARTICLE 12 - GENERAL PROVISIONS

12.1 Interpretation and Intent

- **12.1.1** This Agreement, Plans, Standard Specifications and Details, Special Provisions, Performance Bond, Payment Bond, Certificates of Insurance, and Change Orders (if any) are by this reference made a part of this Agreement to the same extent as if set forth herein in full.
- **12.1.2** The Contract Documents are intended to permit the parties to complete the Work and all obligations required by the Contract Documents within the Contract Times for the Contract Price. The Contract Documents are intended to be complementary and interpreted in harmony so as to avoid conflict, with words and phrases interpreted in a manner consistent with construction and design industry standards.
- **12.1.3** In the event of any inconsistency, conflict, or ambiguity between or among the Contract Documents, the Contract Documents shall take precedence in the order in which they are listed in the definition of Contract Documents in Article 1. In the event conflicts occur between the drawings and specifications, the Design-Builder is deemed to have estimated the more expensive method unless he has asked for and received a written decision from the City determining which method or material will be required.
- **12.1.4** The Contract Documents form the entire agreement between the City and the Design-Builder and by incorporation herein are as fully binding on the parties as if repeated herein. No oral

representations or other agreements have been made by the parties except as specifically stated in the Contract Documents.

- **12.2 Time is of the Essence.** The City and the Design-Builder mutually agree that time is of the essence with respect to the dates and times set forth in the Contract Documents.
- **12.3 Mutual Obligations.** The City and the Design-Builder commit at all times to cooperate fully with each other, and proceed on the basis of trust and good faith, to permit each party to realize the benefits afforded under the Contract Documents.

12.4 Work Product

- **12.4.1** All work products (electronically or manually generated) including but not limited to: cost estimates, studies, design analyses, original mylar drawings, Computer Aided Drafting and Design (CADD) file diskettes, and other related documents which are prepared or procured in the performance of this Agreement (collectively referred to as documents) are to be and remain the property of the City and are to be delivered to the City before the final payment is made to the Design-Builder or pursuant to Section 9.2.1. In the event these documents are altered, modified or adapted without the written consent of the Design-Builder or the Subconsultants, which consent the Design-Builder or the Subconsultants shall not unreasonably withhold, the City agrees to hold the Design-Builder and the Subconsultants harmless to the extent permitted by law from the legal liability arising out of the City's alteration, modification or adaption of the documents.
- **12.4.2** The copyrights, patents, trade secrets or other intellectual property rights associated with the ideas, concepts, techniques, inventions, processes or works of authorship developed, created by the Design-Builder, its Subconsultants or personnel, during the course of performing this Agreement or arising out of the Project shall belong to the Design-Builder.
- **12.4.3** With this Agreement, the Design-Builder and its Subconsultants hereby grant a license to the City, its agents, employees, and representatives for an indefinite period of time to reasonably use, make copies, and distribute as appropriate the documents, works or deliverables developed or created as a result of the Project and this Agreement. This license also includes the making of derivative works. In the event that the derivative works require the City to alter or modify the documents, then the provisions of Section 12.4.1 apply.
- **12.5** Assignment. Neither the Design-Builder nor the City shall, without the written consent of the other assign, transfer or sublet any portion or part of the Work or the obligations required by the Contract Documents.
- **12.6 Successorship.** The Design-Builder and the City intend that the provisions of the Contract Documents are binding upon the parties, their employees, agents, heirs, successors and assigns.
- **12.7 Third Party Beneficiary.** Nothing under this Agreement shall be construed to give any rights or benefits in the Agreement to anyone other than the City and the Design-Builder, and all duties and responsibilities undertaken pursuant to this Agreement will be for the sole and exclusive benefit of the City and the Design-Builder and not for the benefit of any other party.
- **12.8 Governing Law.** The Agreement and all Contract Documents shall be deemed to be made under, and shall be construed in accordance with and governed by the laws of the State of Arizona without regard to the conflicts or choice of law provisions thereof. Any action to enforce any provision of this Agreement or to obtain any remedy with respect hereto shall be brought in the Superior Court, Pinal County, Arizona, and for this purpose, each party hereby expressly and irrevocably consents to the jurisdiction and venue of such Court.
- **12.9** Severability. If any provision of this Agreement or the application thereof to any person or circumstance shall be invalid, illegal or unenforceable to any extent, the remainder of this

Agreement and the application thereof shall not be affected and shall be enforceable to the fullest extent permitted by law.

- **12.10 No Waiver.** The failure of either party to enforce any of the provisions of this Agreement or to require performance of the other party of any of the provisions hereof shall not be construed to be a waiver of such provisions, nor shall it affect the validity of this Agreement or any part thereof, or the right of either party to thereafter enforce each and every provision.
- **12.11 Headings.** The headings used in this Agreement, or any other Contract Documents, are for ease of reference only and shall not in any way be construed to limit or alter the meaning of any provision.

12.12 Notice

12.12.1 Unless otherwise provided, any notice, request, instruction or other document to be given under this Agreement by any party to any other party shall be in writing and shall be delivered in person or by courier or facsimile transmission or mailed by certified mail, postage prepaid, return receipt requested and shall be deemed given upon (a) confirmation of receipt of a facsimile transmission, (b) confirmed delivery by hand or standard overnight mail or (c) upon the expiration of three (3) business days after the day mailed by certified mail, as follows:

To Design-Builder:	Fritz Behrhorst, Vice President of Pre-Construction <u>FBehrhorst@Haydonbc.com</u> Or Les Keeble, Vice President of Operations <u>LKeeble@Haydonbc.com</u> Haydon Building Corp 4640 E Cotton Gin Loop Phoenix, AZ 85040 602-296-1496
to City:	Brent Billingsley, Development Services Director City of Maricopa, Development Services Department 45145 W. Madison Ave Maricopa, Arizona 85139 (520)316-6942 Phone (520)568-0138 Fax
With a Copies to:	Adam Brill, City's Program Manager ABACUS Project Management 3030 N. Central Ave, Suite 1207 Phoenix, AZ 85012 (602)282-8860 Phone
And:	Pattie LaCombe, Purchasing Manager City of Maricopa, Financial Services Department 45145 W. Madison Ave Maricopa, Arizona 85139 (520)316-6846 Phone (520)316-6959 Fax

or to such other place and with such other copies as either Party may designate as to itself by written notice to the other Party. Rejection, any refusal to accept or the inability to deliver because of changed address of which no notice was given shall be deemed to be receipt of the notice as of the date of such rejection, refusal or inability to deliver.

12.12.2 <u>Notices Related to Payment, Securities-in-lieu, Bonds</u>. Any notice, request, instruction or other document to be given under this Agreement by any party to any other party related to payment, securities-in-lieu, bonds or other instrument securing the performance of this Agreement, including but not limited to, bid bonds, performance bonds, payment bonds or letters of credit, shall be in writing and shall be delivered in person or by courier or facsimile transmission or mailed by certified mail, postage prepaid, return receipt requested and shall be delivery by hand or standard overnight mail or (c) upon the expiration of three (3) business days after the day mailed by certified mail, as follows:</u>

To Design-Builder: Fritz Behrhorst, Vice President of Pre-Construction FBehrhorst@Havdonbc.com Or Les Keeble, Vice President of Operations LKeeble@Haydonbc.com Haydon Building Corp 4640 E Cotton Gin Loop Phoenix, AZ 85040 602-296-1496 to City: Brent Billingsley, Development Services Director City of Maricopa, Development Services Department 45145 W. Madison Ave Maricopa, Arizona 85139 (520)316-6942 Phone (520)568-0138 Fax With a Copies to: Adam Brill, City's Program Manager **ABACUS Project Management** 3030 N. Central Ave, Suite 1207 Phoenix, AZ 85012 (602)282-8860 Phone And: Pattie LaCombe, Purchasing Manager City of Maricopa, Financial Services Department 45145 W. Madison Ave Maricopa, Arizona 85139 (520)316-6846 Phone (520)316-6859 Fax

or to such other place and with such other copies as either Party may designate as to itself by written notice to the other Party. Rejection, any refusal to accept or the inability to deliver because of changed address of which no notice was given shall be deemed to be receipt of the notice as of the date of such rejection, refusal or inability to deliver.

12.13 Amendments. The Contract Documents may not be changed, altered, or amended in any way except in writing signed by a duly authorized representative of each party.

12.14 Requirements of Design-Builder

12.14.1 Any Design-Builder in performing under this Contract shall not discriminate against any worker, employee or applicant, or any member of the public, because of race, color, religion, sex, national origin, age, or disability nor otherwise commit an unfair employment practice. The Design-Builder shall ensure that applicants are employed, and employees are dealt with during employment without regard to their race, color, religion, sex, national origin, age, or disability.

Such action shall include but not be limited to the following: Employment, promotion, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rate of pay or other forms of compensation; and selection for training; including apprenticeship. The Design-Builder further agrees that this clause will be incorporated in all subcontracts with all labor organizations furnishing skilled, unskilled and union labor, or who may perform any such labor or services in connection with this contract.

- **12.14.2** The City of Maricopa extends to each individual, firm, vendor, supplier, contractor, and subcontractor an equal economic opportunity to compete for City business and strongly encourages voluntary utilization of Disadvantaged and/or Small-owned business to reflect both the industry and community ethnic composition.
- **12.15 Compliance with Federal Laws.** The Design-Builder understands and acknowledges the applicability of the American with Disabilities Act, the Immigration Reform and Control Act of 1986 and the Drug Free Workplace Act of 1989 to it. The Design-Builder agrees to comply with these laws in performing this Agreement and to permit the City to verify such compliance.
- **12.16** Business with Sudan and Iran. Pursuant to A.R.S. §§ 35-391.06 and 35-393.06, the Design-Builder certifies that it does not have a scrutinized business operation, as defined in A.R.S. §§ 35-391 and 35-393, in either Sudan or Iran.
- **12.17** Legal Worker Requirements. The City of Maricopa is prohibited by A.R.S. § 41-4401 from awarding a contract to any Design-Builder who fails, or whose Subcontractors fail, to comply with A.R.S. § 23-214(A). Therefore, the Design-Builder agrees that:
- **12.17.1** The Design-Builder and each Subcontractor it uses warrants their compliance with all federal immigration laws and regulations that relate to their employees and their compliance with A.R.S. § 23-214, subsection A.
- **12.17.2** A breach of a warranty under paragraph 12.17.1 shall be deemed a material breach of the Contract that is subject to penalties up to and including termination of the Contract.
- **12.17.3** The City of Maricopa retains the legal right to inspect the papers of any Design-Builder or Subcontractor employee who works on the Contract to ensure that the Design-Builder or Subcontractor is complying with the warranty under paragraph 12.17.1.

12.18 Lawful Presence Requirement

Pursuant to A.R.S. §§ 1-501 and 1-502, the City of Maricopa is prohibited from awarding a contract to any natural person who cannot establish that such person is lawfully present in the United States. To establish lawful presence, a person must produce qualifying identification and sign a City-provided affidavit affirming that the identification provided is genuine. This requirement will be imposed at the time of contract award. This requirement does not apply to business organizations such as corporations, partnerships or limited liability companies.

- **12.19** Independent Contractor. The Design-Builder is and shall be an independent contractor. Any provisions in this Agreement that may appear to give the City the right to direct the Design-Builder as to the details of accomplishing the Work or to exercise a measure of control over the Work means that the Design-Builder shall follow the wishes of the City as to the results of the Work only.
- **12.20 City's Right Of Cancellation.** All parties hereto acknowledge that this Agreement is subject to cancellation by the City of Maricopa pursuant to the provisions of Section 38-511, Arizona Revised Statutes.

12.21 Data Confidentiality and Data Security

12.21.1 Data Confidentiality. As used in the Contract, "data" means all information, whether written or verbal, including plans, photographs, studies, investigations, audits, analyses, samples, reports, calculations, internal memos, meeting minutes, data field notes, work product, proposals,

correspondence and any other similar documents or information prepared by, obtained by, or transmitted to the Design-Builder or its Subconsultants/Subcontractors in the performance of this Contract.

- **12.12.1.1** The parties agree that all data, regardless of form, including originals, images, and reproductions, prepared by, obtained by, or transmitted to the Design-Builder or its Subconsultants/Subcontractors in connection with the Design-Builder's or its Subconsultant's/Subcontractor's performance of this Contract is confidential and proprietary information belonging to the City.
- **12.21.1.2** Except as specifically provided in this Contract, the Design-Builder or its Subconsultants/Subcontractors shall not divulge data to any third party without prior written consent of the City. The Design-Builder or its Subconsultants/Subcontractors shall not use the data for any purposes except to perform the services required under this Contract. These prohibitions shall not apply to the following data provided the Design-Builder or its Subconsultants/Subcontractors have first given the required notice to the City:
- **12.21.1.2.1** Data which was known to the Design-Builder or its Subconsultants/Subcontractors prior to its performance under this Contract unless such data was acquired in connection with work performed for the City;
- **12.21.1.2.2** Data which was acquired by the Design-Builder or its Subconsultants/Subcontractors in its performance under this Contract and which was disclosed to the Design-Builder or its Subconsultants/Subcontractors by a third party, who to the best of the Design-Builder's or its Subconsultant's/Subcontractor's knowledge and belief, had the legal right to make such disclosure and the Design-Builder or its Subconsultants/Subcontractors are not otherwise required to hold such data in confidence; or
- **12.21.1.2.3** Data which is required to be disclosed by virtue of law, regulation, or court order, to which the Design-Builder or its Subconsultant's/Subcontractor's are subject.
- **12.21.1.3** In the event the Design-Builder or its Subconsultants/Subcontractors are required or requested to disclose data to a third party, or any other information to which the Design-Builder or its Subconsultants/Subcontractors became privy as a result of any other contract with the City, the Design-Builder shall first notify the City as set forth in this section of the request or demand for the data. The Design-Builder or its Subconsultants/Subcon
- **12.21.1.4** The Design-Builder, unless prohibited by law, within ten (10) days after completion of services for a third party on real or personal property owned or leased by the City, the Design-Builder or its Subconsultants/Subcontractors shall promptly deliver, as set forth in this section, a copy of all data to the City. All data shall continue to be subject to the confidentiality agreements of this Contract.
- **12.21.1.5** The Design-Builder or its Subconsultants/Subcontractors assume all liability for maintaining the confidentiality of the data in its possession and agrees to compensate the City if any of the provisions of this section are violated by the Design-Builder, its employees, agents or Subconsultants/Subcontractors. Solely for the purposes of seeking injunctive relief, it is agreed that a breach of this section shall be deemed to cause irreparable harm that justifies injunctive relief in court. The Design-Builder agrees that the requirements of this Section shall be incorporated into all subcontracts entered into by the Design-Builder. A violation of this Section may result in immediate termination of this Contract without notice.
- **12.21.2** Personal Identifying Information-Data Security. Personal identifying information, financial account information, or restricted City information, whether electronic format or hard copy, must be secured and protected at all times. At a minimum, the Design-Builder must encrypt and/or password protect electronic files. This includes data saved to laptop computers, computerized devices or removable storage devices.

- **12.21.2.1** When personal identifying information, financial account information, or restricted City information, regardless of its format, is no longer necessary, the information must be redacted or destroyed through appropriate and secure methods that ensure the information cannot be viewed, accessed, or reconstructed.
- **12.21.2.2** In the event that data collected or obtained by the Design-Builder or its Subconsultants/Subcontractors in connection with this Contract is believed to have been compromised, the Design-Builder or its Subconsultants/Subcontractors shall immediately notify the Project Manager and the City Senior Representative. The Design-Builder agrees to reimburse the City for any costs incurred by the City to investigate potential breaches of this data and, where applicable, the cost of notifying individuals who may be impacted by the breach.
- **12.21.2.3** The Design-Builder agrees that the requirements of this Section shall be incorporated into all subcontracts entered into by the Design-Builder. It is further agreed that a violation of this Section shall be deemed to cause irreparable harm that justifies injunctive relief in court. A violation of this Section may result in immediate termination of this Contract without notice.
- **12.21.2.4** The obligations of the Design-Builder or its Subconsultants/Subcontractors under this Section shall survive the termination of this Contract.

12.24 Conflict Of Interest

- **12.24.1** To evaluate and avoid potential conflicts of interest, the Design-Builder shall provide written notice to the City, as set forth in this Section, of any work or services performed by the Design-Builder for third parties that may involve or be associated with any real property or personal property owned or leased by the City. Such notice shall be given seven (7) business days prior to commencement of the Project by the Design-Builder for a third party, or seven (7) business days prior to an adverse action as defined below. Written notice and disclosure shall be sent to the City Senior Representative identified in Subsection 8.4.1.1.
- **12.24.2** Actions that are considered to be adverse to the City under this Agreement include but are not limited to:
- **12.24.2.1** Using data as defined in this Agreement acquired in connection with this Agreement to assist a third party in pursuing administrative or judicial action against the City;
- **12.24.2.2** Testifying or providing evidence on behalf of any person in connection with an administrative or judicial action against the City; and
- **12.24.2.3** Using data to produce income for the Design-Builder or its employees independently of performing the services under this Agreement, without the prior written consent of the City.
- **12.24.3** The Design-Builder represents that except for those persons, entities and projects identified to the City, the services to be performed by the Design-Builder under this Agreement are not expected to create an interest with any person, entity, or third party project that is or may be adverse to the interests of the City.
- 12.24.4 The Design-Builder's failure to provide a written notice and disclosure of the information as set forth in this Section shall constitute a material breach of this Agreement.
- **12.25** Legal Requirements. Design-Builder shall perform the Work in accordance with all Legal Requirements and shall provide all notices applicable to the Work as required by the Legal Requirements.
- **12.26** Confidentiality of Plans. Any plans, records, documents, shop drawings, other plans or drawing, or specifications the Design-Builder generates regarding this Project are for official use

only and shall include the following language: "These plans are for official use only and may not be shared with others except as required to fulfill the obligations of the Design-Builder's contract with the City of Maricopa."

12.27 Hazardous Materials

- **12.27.1** Unless included in the Work, if the Design-Builder encounters onsite or as material to be incorporated in the Work any material which he reasonably believes to contain asbestos, polychlorinated biphenyl (PCB), or other hazardous substances or materials regulated by public health laws, he shall immediately stop work and report the condition to the City.
- **12.27.2** If the material is found to contain asbestos, PCB or other hazardous substances or materials regulated by public health laws, the Design-Builder shall not resume work in the affected area until the material has been abated or rendered harmless. The Design-Builder and the City may agree, in writing, to continue work in non-affected areas onsite.
- **12.27.3** An extension of Contract Time may be granted in accordance with Article 6.
- **12.27.4** The Design-Builder will comply with all applicable laws/ordinances and regulations and take all appropriate health and safety precautions upon discovery.
- **12.27.5** Notwithstanding the preceding provisions of this Section 12.27, the City is not responsible for Hazardous Conditions introduced to the Site by the Design-Builder, Subcontractors or anyone for whose acts they may be liable. The Design-Builder shall indemnify, defend and hold harmless Owner and Owner's officers, directors, employees and agents from and against all claims, losses, damages, liabilities and expenses, including attorneys' fees and expenses, arising out of or resulting from those Hazardous Conditions introduced to the Site by the Design-Builder, Subcontractors or anyone for whose acts they may be liable.
- **12.28 Traffic Control.** The Design-Builder will comply with all provisions of the Manual on Uniform Traffic Control Devices (MUTCD).
- **12.29 Covenant Against Contingent Fees.** The Design-Builder warrants that no person has been employed or retained to solicit or secure this Agreement upon an agreement or understanding for a commission, percentage, brokerage, or contingent fee, and that no member of the City Council, or any employee of the City of Maricopa has any interest, financially, or otherwise, in the Design-Build firm. For breach or violation of this warranty, the City of Maricopa shall have the right to annul this Agreement without liability, or at its discretion to deduct from the Compensation or consideration, the full amount of such commission, percentage, brokerage, or contingent fee.
- **12.30** Fair Treatment of Workers. The Design-Builder shall keep fully informed of all Federal and State laws, County and City ordinances, regulations, codes and all orders and decrees of bodies or tribunals having jurisdiction or authority, which in any way affects the conduct of Work. The Design-Builder shall at all times observe and comply with all such laws, ordinances, regulations, codes. orders and decrees; this includes but is not limited to laws and regulations ensuring fair and equal treatment for all employees against unfair employment practices, including OSHA and the Fair Labor Standards Act (FLSA). The Design-Builder shall protect and indemnify the City and its representatives against any claim or liability arising from or based on the violation of such, whether by himself or his employees.
- **12.31** All Work performed shall conform to all applicable City of Maricopa codes, ordinances and requirements as outlined in this Agreement. If there is a conflict in interpretation between provisions in this Agreement and those in Exhibit "A", the provisions in this Agreement shall prevail.
- **12.32** The following exhibits are included in this Agreement and are incorporated by reference:

Exhibit A - Project Description Exhibit B – Request for Statement of Qualifications and Design-Builder's Response Exhibit C - Design/Preconstruction Scope & Fee Exhibit D - Submittal Requirements for the GMP

This Agreement shall be in full force and effect only when it has been approved and executed by the duly authorized City officials.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement on _____

CITY OF MARICOPA, ARIZONA, an Arizona Municipal Corporation

Haydon Building Corp

By:___

Anthony Smith

By:_____ Fritz Behrhorst,

nony Smiti

Vice President of Pre-Construction

ATTEST:

Mayor

City Clerk

APPROVED AS TO FORM:

City Attorney

.